UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): March 29, 2023

Baudax Bio, Inc.

(Exact name of registrant as specified in its charter)

Pennsylvania (State or other jurisdiction of incorporation or organization) 001-39101 (Commission File Number) 47-4639500 (I.R.S. Employer Identification No.)

490 Lapp Road, Malvern, Pennsylvania (Address of principal executive offices)

19355 (Zip Code)

Registrant's telephone number, including area code: (484) 395-2470

Not Applicable (Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the

	Common Stock, par value \$0.01	BXRX	Nasdaq Capital Market					
	Title of Each Class	Trading Symbol	Name of Exchange on Which Registered					
Securities registered pursuant to Section 12(b) of the Act:								
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))							
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))							
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)							
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)							
	<i>7</i>							

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01 Entry into a Material Definitive Agreement.

Amendment No. 5 to Credit Agreement

On March 29, 2023 (the "Closing Date"), Baudax Bio, Inc. (the "Company") entered into that certain Amendment No. 5 and Consent to Credit Agreement (the "Amendment") by and among the Company, Baudax Bio N.A. LLC ("Baudax LLC"), Baudax Bio Limited, Wilmington Trust, National Association, solely in its capacity as administrative and collateral agent (the "Agent") and the lenders party thereto (the "Lenders"). The Amendment amends that certain Credit Agreement, dated as of May 29, 2020, by and among the Company, the Agent, and the Lenders (as amended, the "Credit Agreement").

Pursuant to the terms of the Amendment, the Lenders consented to the transactions contemplated by the Transfer Agreement (as defined below) and agreed to release and discharge any liens granted or held by the Lenders in respect of the Assets (as defined below). Pursuant to the terms of the Transfer Agreement, the parties also agreed to, among other things, amend the minimum liquidity covenants under the Credit Agreement to require that the Company maintains \$2.5 million of liquidity at all times.

Warrants

In connection with the Amendment, on the Closing Date, the Company issued warrants to MAM Eagle Lender, LLC ("MAM") to purchase an aggregate of 785,026 shares of the Company's common stock, par value \$0.01 per share (the "Common Stock") at an exercise price equal to \$1.8951 per share (the "Warrants"). The Warrants are exercisable until the tenth anniversary of the Closing Date. The holder of each Warrant has the right to net exercise the Warrant for shares of Common Stock upon exercise. A holder (together with its affiliates) may not exercise any portion of the Warrant to the extent that the holder would own more than 9.99% of the Company's outstanding Common Stock immediately after exercise. However, upon at least 61 days' prior notice from the holder to the Company, a holder with a 9.99% ownership blocker may increase or decrease the maximum amount of ownership of outstanding Common Stock after exercising the holder's Warrant to any other percentage not in excess of 9.99% of the number of the Company's Common Stock outstanding immediately after giving effect to the exercise, as such percentage ownership is determined in accordance with the terms of the Warrants

In connection with the Warrants, the Company also granted registration rights to holders of the Warrants to register the common stock subject to such Warrants (the "Warrant Shares") under the Securities Act of 1933, as amended (the "Securities Act") in the event the Company files a registration statement with the U.S. Securities and Exchange Commission under the Securities Act covering its equity securities, subject to the terms and conditions included in the registration rights agreement by and between the Company and MAM Eagle Lender, LLC, entered into on March 29, 2023 (the "Registration Rights Agreement").

The Company relied on the exemption from registration contained in Section 4(2) of the Securities Act, and Regulation D, Rule 506 thereunder, for the issuance of the Warrants and the Warrant Shares. As part of executing the Amendment and receiving the Warrants and the Warrant Shares, each of lenders under the credit agreement represented that it is an "accredited investor" as defined in Regulation D of the Securities Act and that the securities purchased by it will be acquired solely for its own account for investment and not with a view to or for sale or distribution of the Warrants or the Warrant Shares or any part thereof.

The foregoing summary of the Amendment and the Registration Rights Agreement do not purport to be complete and are qualified in their entirety by reference to the Amendment and the Registration Rights Agreement, copies of which are filed as Exhibit 10.1 and 10.2 hereto and are incorporated herein by reference.

Alkermes Agreement

On March 29, 2023, the Company entered into an Asset Transfer Agreement with Alkermes Pharma Ireland Limited ("Alkermes") (the "Transfer Agreement"). Under the terms of the Transfer Agreement, the Company transferred the rights to certain patents, trademarks, equipment, data and other rights related to ANJESO® (the "Assets") to Alkermes. The Company also is withdrawing the New Drug Application ("NDA") related to ANJESO and agreed, if elected by Alkermes at a later date, to transfer such withdrawn NDA to Alkermes at no additional cost.

Additionally, under the Transfer Agreement, the Company granted Alkermes a non-exclusive, perpetual and irrevocable, royalty-free and fully paid-up worldwide license, to the additional intellectual property owned by the Company necessary to or useful to exploit ANJESO[®]. In consideration of the transfer of the Assets, the parties agreed to the termination of (i) the Purchase and Sale Agreement, dated March 7, 2015 by and among Alkermes, the Company and the other parties thereto (as amended, the "PSA"), (ii) the Asset Transfer and License Agreement, dated April 10, 2015 by and among Alkermes, the Company and the other parties thereto (as amended, the "ATLA"); and (iii) the Development, Manufacturing and Supply Agreement, dated as of July 10, 2015 by and between the Company and Alkermes (as amended, the "Manufacturing Agreement") between the parties related to ANJESO (the PSA, ATLA and Manufacturing Agreement, collectively, the "ANJESO Agreements"). In connection with the termination of the ANJESO Agreements, no further payments of any kind pursuant to the ANJESO Agreements will be payable by the Company to Alkermes.

The foregoing description of the Transfer Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Transfer Agreement, which will be filed with the Company's upcoming Form 10-Q for the quarter ended March 31, 2023.

Item 1.02 Termination of a Material Definitive Agreement.

The disclosure set forth in Item 1.01 of this Current Report on Form8-K under the heading "Alkermes Agreement" is hereby incorporated by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The disclosure set forth in Item 1.01 of this Current Report on Form8-K under the heading "Alkermes Agreement" is hereby incorporated by reference.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The disclosure set forth in Item 1.01 of this Current Report on Form8-K under the heading "Amendment No. 5 to Credit Agreement" is hereby incorporated by reference.

Item 3.02 Unregistered Sale of Equity Securities.

The disclosure set forth in Item 1.01 of this Current Report on Form8-K under the heading "Warrants" is hereby incorporated by reference.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

The following exhibits are being filed herewith:

Exhibit No.	Document
4.1	Common Stock Purchase Warrant, dated March 29, 2023, in favor of MAM Eagle Lender, LLC.
10.1	Amendment No. 5 to Credit Agreement, dated March 29, 2023, by and among Baudax Bio, Inc., Baudax Bio N.A. LLC, Baudax Bio Limited, Wilmington Trust, National Association, and the Lenders party thereto.
10.2	Registration Rights Agreement between the Baudax Bio, Inc. and MAM Eagle Lender, LLC, dated March 29, 2023.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Baudax Bio, Inc.

/s/ Gerri A. Henwood By:

Name: Gerri A. Henwood
Title: President and Chief Executive Officer

Date: March 31, 2023

THIS WARRANT AND THE SECURITIES PURCHASABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR UNDER THE SECURITIES LAWS OF ANY STATE AND MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED UNDER SAID ACT AND ANY APPLICABLE STATE SECURITIES LAWS, UNLESS AN EXEMPTION FROM SUCH REGISTRATION IS AVAILABLE.

BAUDAX BIO, INC.

WARRANT TO PURCHASE COMMON STOCK

Warrant No.: 2

Number of Shares of Common Stock: 785,026 Date of Issuance: March 29, 2023 (**'Issuance Date'**)

Baudax Bio, Inc., a company organized under the laws of the Commonwealth of Pennsylvania (the "Company"), hereby certifies that, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, MAM EAGLE LENDER, LLC, the registered holder hereof or its permitted assigns (the "Holder"), is entitled, subject to the terms set forth below, to purchase from the Company, at the Exercise Price (as defined below) then in effect, at any time or times on or after the Issuance Date, but not after 11:59 p.m., New York time, on the Expiration Date, (as defined below), 785,026 fully paid nonassessable shares of Common Stock (as defined below), subject to adjustment as provided herein (the "Warrant Shares"). Except as otherwise defined herein, capitalized terms in this Warrant to Purchase Common Stock (including any Warrants to Purchase Common Stock issued in exchange, transfer or replacement hereof, this "Warrant"), shall have the meanings set forth in Section 17. This Warrant is one of the Warrants to purchase Common Stock (the "Warrants") issued pursuant to that certain Credit Agreement, dated as of May 29, 2020, by and among the Company, Wilmington Trust, National Association, as agent, and the lenders party thereto.

1. EXERCISE OF WARRANT.

(a) <u>Mechanics of Exercise</u>. Subject to the terms and conditions hereof (including, without limitation, the limitations set forth in Section 1(f)), this Warrant may be exercised by the Holder at any time or times on or after the Issuance Date, in whole or in part (but not as to fractional shares), by delivery (whether via facsimile, electronic mail or otherwise) of a written notice, in the form attached hereto as <u>Exhibit A</u> (the "Exercise Notice"), of the Holder's election to exercise this Warrant. Within one (1) Trading Day following the delivery of the Exercise Notice, the Holder shall make payment to the Company of an amount equal to the Exercise Price in effect on the date of such exercise multiplied by the number of Warrant Shares as to which this Warrant is being exercised (the "Aggregate Exercise Price") in cash by wire transfer of immediately available funds or if the provisions of Section 1(d) are applicable, by notifying the Company that this Warrant is being exercised pursuant to a Cashless Exercise (as defined in Section 1(d)). The Holder shall not be required to deliver the original Warrant in order to effect an exercise hereunder, nor shall any ink-original signature or medallion guarantee (or other type of guarantee or notarization) with respect to any Exercise Notice be required. Execution and delivery of the Exercise Notice with respect to less than all of the Warrant Shares shall have

the same effect as cancellation of the original Warrant and issuance of a new Warrant evidencing the right to purchase the remaining number of Warrant Shares and the Holder shall not be required to physically surrender this Warrant to the Company until the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company for cancellation within five (5) Trading Days of the date on which the final Exercise Notice has been delivered to the Company. On or before the first (1st) Trading Day following the date on which the Holder has delivered the applicable Exercise Notice, the Company shall transmit by facsimile or electronic mail an acknowledgment of confirmation of receipt of the Exercise Notice, in the form attached to the Exercise Notice, to the Holder and the Company's transfer agent (the "Transfer Agent"). So long as the Holder delivers the Aggregate Exercise Price (or notice of a Cashless Exercise) on or prior to the first (1st) Trading Day following the date on which the Exercise Notice has been delivered to the Company, then on or prior to the earlier of (i) the second (2nd) Trading Day and (ii) the number of Trading Days comprising the Standard Settlement Period, in each case following the date on which the Exercise Notice has been delivered to the Company, or, if the Holder does not deliver the Aggregate Exercise Price (or notice of a Cashless Exercise) on or prior to the first (1st) Trading Day following the date on which the Exercise Notice has been delivered to the Company, then on or prior to the first (1st) Trading Day following the date on which the Aggregate Exercise Price (or notice of a Cashless Exercise) is delivered (such earlier date, the "Share Delivery Date"), the Company shall (X) provided that the Transfer Agent is participating in The Depository Trust Company ("DTC") Fast Automated Securities Transfer Program, credit such aggregate number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the Holder's or its designee's balance account with DTC through its Deposit / Withdrawal At Custodian system, or (Y) if the Transfer Agent is not participating in the DTC Fast Automated Securities Transfer Program, issue and dispatch by overnight courier to the address as specified in the Exercise Notice, a certificate, registered in the Company's share register in the name of the Holder or its designee, for the number of Warrant Shares to which the Holder is entitled pursuant to such exercise. The Company shall be responsible for all fees and expenses of the Transfer Agent and all fees and expenses with respect to the issuance of Warrant Shares via DTC, if any, including without limitation for same day processing. Upon delivery of the Exercise Notice, the Holder shall be deemed for all corporate purposes to have become the holder of record and beneficial owner of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date such Warrant Shares are credited to the Holder's DTC account or the date of delivery of the certificates evidencing such Warrant Shares, as the case may be. If this Warrant is physically delivered to the Company in connection with any exercise pursuant to this Section 1(a) and the number of Warrant Shares represented by this Warrant submitted for exercise is greater than the number of Warrant Shares being acquired upon an exercise, then the Company shall as soon as practicable and in no event later than three (3) Trading Days after any exercise and at its own expense, issue and deliver to the Holder (or its designee) a new Warrant (in accordance with Section 7(d)) representing the right to purchase the number of Warrant Shares issuable immediately prior to such exercise under this Warrant, less the number of Warrant Shares with respect to which this Warrant is exercised. No fractional Warrant Shares are to be issued upon the exercise of this Warrant, but rather the number of Warrant Shares to be issued shall be rounded up to the nearest whole number. The Company shall pay any and all transfer, stamp, issuance and similar taxes, costs and expenses (including, without limitation, fees and expenses of the Transfer Agent) which may be payable with respect to the issuance and delivery of Warrant Shares upon exercise of this Warrant. The

Company's obligations to issue and deliver Warrant Shares in accordance with the terms and subject to the conditions hereof are absolute and unconditional, irrespective of any action or inaction by the Holder to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any Person or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination; provided, however, that the Company shall not be required to deliver Warrant Shares with respect to an exercise prior to the Holder's delivery of the Aggregate Exercise Price (or notice of a Cashless Exercise) with respect to such exercise.

(b) Exercise Price. For purposes of this Warrant, "Exercise Price" means \$1.8951 per share, subject to adjustment as provided herein.

(c) Company's Failure to Timely Deliver Securities. If the Company shall fail for any reason or for no reason to issue to the Holder on or prior to the applicable Share Delivery Date, if (x) the Transfer Agent is not participating in the DTC Fast Automated Securities Transfer Program, a certificate for the number of Warrant Shares that are the subject of the Exercise Notice (the "Exercise Notice Warrant Shares") to which the Holder is entitled and register such Exercise Notice Warrant Shares on the Company's share register or (y) the Transfer Agent is participating in the DTC Fast Automated Securities Transfer Program, to credit the Holder's balance account with DTC, for such Exercise Notice Warrant Shares to which the Holder is entitled upon the Holder's exercise of this Warrant (an "Exercise Failure"), then, in addition to all other remedies available to the Holder, if on or prior to the applicable Share Delivery Date if the Transfer Agent is not participating in the DTC Fast Automated Securities Transfer Program, the Company shall fail to issue and deliver a certificate to the Holder and register such Exercise Notice Warrant Shares on the Company's share register or, if the Transfer Agent is participating in the DTC Fast Automated Securities Transfer Program, credit the Holder's balance account with DTC for the number of Exercise Notice Warrant Shares to which the Holder is entitled upon the Holder's exercise hereunder or pursuant to the Company's obligation pursuant to clause (ii) below, and if on or after such Trading Day the Holder is required by its brokerage firm to purchase (in an open market transaction or otherwise) or the Holder's brokerage firm otherwise purchases Common Stock to deliver in satisfaction of a sale by the Holder of shares of Common Stock issuable upon such exercise that the Holder anticipated receiving from the Company (a "Buy-In"), then the Company shall, within three (3) Trading Days after the Holder's written request, (A) pay in cash to the Holder the amount, if any, by which (x) the Holder's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased exceeds (y) the amount obtained by multiplying (1) the number of Warrant Shares that the Company was required to deliver to the Holder in connection with the exercise at issue time by (2) the price at which the sell order giving rise to such purchase obligation was executed, and (B) at the option of the Holder, either reinstate the portion of the Warrant and equivalent number of shares of Common Stock for which such exercise was not honored (in which case such exercise shall be deemed rescinded) or deliver to the Holder the number of shares of Common Stock that would have been issued had the Company timely complied with its exercise and delivery obligations hereunder. For example, if the Holder purchases Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted exercise of shares of Common Stock with an aggregate sale price giving rise to such purchase obligation of \$10,000, under clause (A) of the immediately preceding sentence the Company shall be required to pay the Holder \$1,000. The Holder shall provide the Company written notice indicating the amounts payable to the Holder with respect of the BuyIn and, upon the request of the Company, evidence of the amount of such loss. Nothing herein shall limit the Holder's right to pursue any other remedies available to it hereunder, at law or in equity, including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver certificates representing Warrant Shares (or to electronically deliver such Warrant Shares) upon the exercise of this Warrant as required pursuant to the terms hereof. While this Warrant is outstanding, the Company shall cause its transfer agent to participate in the DTC Fast Automated Securities Transfer Program. In addition to the foregoing rights, if the Company fails to deliver the applicable number of Warrant Shares upon an exercise pursuant to Section 1 by the applicable Share Delivery Date, then the Holder shall have the right to rescind such exercise in whole or in part and retain and/or have the Company return, as the case may be, any portion of this Warrant that has not been exercised pursuant to such Exercise Notice; provided that the rescission of an exercise shall not affect the Company's obligation to make any payments that have accrued prior to the date of such notice pursuant to this Section 1(c) or otherwise.

(d) <u>Cashless Exercise</u>. The Holder may, in its sole discretion, exercise this Warrant in whole or in part and, in lieu of making the cash payment otherwise contemplated to be made to the Company upon such exercise in payment of the Aggregate Exercise Price, elect instead to receive upon such exercise the "Net Number" of shares of Common Stock determined according to the following formula (a "Cashless Exercise"):

Net Number =
$$(A \times B) - (A \times C)$$

For purposes of the foregoing formula:

A= the total number of shares with respect to which this Warrant is then being exercised.

B= as applicable: (i) the Closing Sale Price of the Common Stock on the Trading Day immediately preceding the date of the applicable Exercise Notice if such Exercise Notice is (1) both executed and delivered pursuant to Section 1(d) hereof on a day that is not a Trading Day or (2) both executed and delivered pursuant to Section 1(a) hereof on a Trading Day prior to the opening of "regular trading hours" (as defined in Rule 600(b)(68) of Regulation NMS promulgated under the federal securities laws) on such Trading Day, (ii) at the option of the Holder, either (x) the Weighted Average Price on the Trading Day immediately preceding the date of the applicable Exercise Notice or (y) the Bid Price of the Common Stock as of the time of the Holder's execution of the applicable Exercise Notice if such Exercise Notice is executed during "regular trading hours" on a Trading Day and is delivered within two (2) hours thereafter pursuant to Section 1(a) hereof (including until two (2) hours after the close of "regular trading hours" on a Trading Day), or (iii) the Closing Sale Price of the Common Stock on the date of the applicable Exercise Notice if the date of such Exercise Notice is a Trading Day and such Exercise Notice is both executed and delivered pursuant to Section 1(a) hereof after the close of "regular trading hours" on such Trading Day.

C= the Exercise Price then in effect for the applicable Warrant Shares at the time of such exercise.

If Warrant Shares are issued in such a cashless exercise, the Company acknowledges and agrees that in accordance with Section 3(a)(9) of the Securities Act of 1933, as amended, the Warrant Shares shall take on the characteristics of the portion of this Warrant being exercised, and the holding period of the portion of this Warrant being exercised may be tacked on to the holding period of the Warrant Shares. The Company agrees not to take any position contrary to this Section 1(d).

(e) <u>Disputes</u>. In the case of a dispute as to the determination of the Exercise Price or the arithmetic calculation of the Warrant Shares, the Company shall promptly issue to the Holder the number of Warrant Shares that are not disputed and resolve such dispute in accordance with Section 11.

(f) Beneficial Ownership. Notwithstanding anything to the contrary contained herein, the Company shall not effect the exercise of any portion of this Warrant, and the Holder shall not have the right to exercise any portion of this Warrant, pursuant to the terms and conditions of this Warrant and any such exercise shall be null and void and treated as if never made, to the extent that after giving effect to such exercise, the Holder together with the other Attribution Parties collectively would beneficially own in excess of 9.99% (the "Maximum Percentage") of the number of shares of Common Stock outstanding immediately after giving effect to such exercise. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by the Holder and the other Attribution Parties shall include the number of shares of Common Stock held by the Holder and all other Attribution Parties plus the number of shares of Common Stock issuable upon exercise of this Warrant with respect to which the determination of such sentence is being made, but shall exclude the number of shares of Common Stock which would be issuable upon (A) exercise of the remaining, unexercised portion of this Warrant beneficially owned by the Holder or any of the other Attribution Parties and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company (including, without limitation, any convertible notes or convertible preferred stock or warrants, including the other Warrants) beneficially owned by the Holder or any other Attribution Party subject to a limitation on conversion or exercise analogous to the limitation contained in this Section 1(f). For purposes of this Section 1(f), beneficial ownership shall be calculated in accordance with Section 13(d) of the Securities Exchange Act of 1934, as amended (the "1934 Act"). For purposes of this Warrant, in determining the number of outstanding shares of Common Stock the Holder may acquire upon the exercise of this Warrant without exceeding the Maximum Percentage, the Holder may rely on the number of outstanding shares of Common Stock as reflected in (x) the Company's most recent Annual Report on Form 10-K, Quarterly Report on Form 10-Q and Current Reports on Form 8-K or other public filing with the Securities and Exchange Commission (the 'SEC"), as the case may be, (y) a more recent public announcement by the Company or (z) any other written notice by the Company or the Transfer Agent setting forth the number of shares of Common Stock outstanding (the "Reported Outstanding Share Number"). If the Company receives an Exercise Notice from the Holder at a time when the actual number of outstanding shares of Common Stock is less than the Reported Outstanding Share Number, the Company shall (i) notify the Holder in writing of the number of shares of Common Stock then outstanding and,

to the extent that such Exercise Notice would otherwise cause the Holder's beneficial ownership, as determined pursuant to this Section 1(f), to exceed the Maximum Percentage, the Holder must notify the Company of a reduced number of Warrant Shares to be purchased pursuant to such Exercise Notice (the number of shares by which such purchase is reduced, the "Reduction Shares") and (ii) as soon as reasonably practicable, the Company shall return to the Holder any exercise price paid by the Holder for the Reduction Shares. For any reason at any time, upon the written or oral request of the Holder, the Company shall within two (2) Business Days confirm orally and in writing or by electronic mail to the Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Company, including this Warrant, by the Holder and any other Attribution Party since the date as of which the Reported Outstanding Share Number was reported. In the event that the issuance of Common Stock to the Holder upon exercise of this Warrant results in the Holder and the other Attribution Parties being deemed to beneficially own, in the aggregate, more than the Maximum Percentage of the number of outstanding shares of Common Stock (as determined under Section 13(d) of the 1934 Act), the number of shares so issued by which the Holder's and the other Attribution Parties' aggregate beneficial ownership exceeds the Maximum Percentage (the "Excess Shares") shall be deemed null and void and shall be cancelled ab initio, and the Holder shall not have the power to vote or to transfer the Excess Shares. As soon as reasonably practicable after the issuance of the Excess Shares has been deemed null and void, the Company shall return to the Holder the exercise price paid by the Holder for the Excess Shares. Upon delivery of a written notice to the Company, the Holder may from time to time increase or decrease the Maximum Percentage to any other percentage not in excess of 9.99% as specified in such notice; provided that (i) any such increase in the Maximum Percentage will not be effective until the sixty-first (61st) day after such notice is delivered to the Company and (ii) any such increase or decrease will apply only to the Holder and the other Attribution Parties and not to any other holder of Warrants that is not an Attribution Party of the Holder. For purposes of clarity, the shares of Common Stock issuable pursuant to the terms of this Warrant in excess of the Maximum Percentage shall not be deemed to be beneficially owned by the Holder for any purpose including for purposes of Section 13(d) or Rule 16a-1(a)(1) of the 1934 Act. No prior inability to exercise this Warrant pursuant to this paragraph shall have any effect on the applicability of the provisions of this paragraph with respect to any subsequent determination of exercisability. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 1(f) to the extent necessary to correct this paragraph or any portion of this paragraph which may be defective or inconsistent with the intended beneficial ownership limitation contained in this Section 1(f) or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitation contained in this paragraph may not be waived and shall apply to a successor holder of this Warrant.

(g) Required Reserve Amount. So long as this Warrant remains outstanding, the Company shall at all times keep reserved for issuance under this Warrant a number of shares of Common Stock at least equal to 100% of the maximum number of shares of Common Stock as shall be necessary to satisfy the Company's obligation to issue shares of Common Stock under the Warrants then outstanding (without regard to any limitations on exercise) (the "Required Reserve Amount"); provided that at no time shall the number of shares of Common Stock reserved pursuant to this Section 1(g) be reduced other than in connection with any exercise of Warrants or

such other event covered by Section 2 below. The Required Reserve Amount (including, without limitation, each increase in the number of shares so reserved) shall be allocated pro rata among the holders of the Warrants based on the number of shares of Common Stock issuable upon exercise of Warrants held by each holder thereof on the Issuance Date (without regard to any limitations on exercise) (the "Authorized Share Allocation"). In the event that a holder shall sell or otherwise transfer any of such holder's Warrants, each transferee shall be allocated a pro rata portion of such holder's Authorized Share Allocation. Any shares of Common Stock reserved and allocated to any Person which ceases to hold any Warrants shall be allocated to the remaining holders of Warrants, pro rata based on the number of shares of Common Stock issuable upon exercise of the Warrants then held by such holders thereof (without regard to any limitations on exercise).

(h) Insufficient Authorized Shares. If at any time while this Warrant remains outstanding the Company does not have a sufficient number of authorized and unreserved shares of Common Stock to satisfy its obligation to reserve for issuance the Required Reserve Amount (an "Authorized Share Failure"), then the Company shall promptly take all action reasonably necessary to increase the Company's authorized shares of Common Stock to an amount sufficient to allow the Company to reserve the Required Reserve Amount for this Warrant then outstanding. Without limiting the generality of the foregoing sentence, as soon as practicable after the date of the occurrence of an Authorized Share Failure, but in no event later than sixty (60) days after the occurrence of such Authorized Share Failure, the Company shall hold a meeting of its stockholders for the approval of an increase in the number of authorized shares of Common Stock. In connection with such meeting, the Company shall provide each stockholder with a proxy statement and shall use its reasonable best efforts to solicit its stockholders' approval of such increase in authorized shares of Common Stock and to cause its board of directors to recommend to the stockholders that they approve such proposal. Notwithstanding the foregoing, if any such time of an Authorized Share Failure, the Company is able to obtain the written consent of a majority of the shares of its issued and outstanding shares of Common Stock to approve the increase in the number of authorized shares of Common Stock, the Company may satisfy this obligation by obtaining such consent and submitting for filing with the SEC an Information Statement on Schedule 14C.

(i) <u>Legend</u>. Each certificate for Warrant Shares issued upon exercise of this Warrant, unless at the time of exercise such Warrant Shares are registered under the Securities Act, shall bear the following legend:

THIS SECURITY HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR UNDER THE SECURITIES LAWS OF ANY STATE AND MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED UNDER SAID ACT AND ANY APPLICABLE STATE SECURITIES LAWS, UNLESS AN EXEMPTION FROM SUCH REGISTRATION IS AVAILABLE.

Any certificate for Warrant Shares issued at any time in exchange or substitution for any certificate bearing such legend (unless at that time such Warrant Shares are registered under the Securities Act) shall also bear such legend unless, in the written opinion of counsel selected by the holder of such certificate (who may be an employee of such holder), which counsel and opinion shall be reasonably acceptable to the Company, the Warrant Shares represented thereby need no longer be subject to restrictions on resale under the Securities Act.

(j) <u>Automatic Cashless Exercise</u>. To the extent that there has not been an exercise by the Holder pursuant to Section 1 hereof, any portion of the Warrant that remains unexercised shall be exercised automatically in whole (not in part), upon the Expiration Date in the manner set forth in Section 1(d).

- 2. <u>ADJUSTMENT UPON SUBDIVISION OR COMBINATION OF COMMON STOCK.</u> If the Company at any time on or after the Issuance Date subdivides (by any stock split, stock dividend, recapitalization or otherwise) one or more classes of its outstanding shares of Common Stock into a greater number of shares, the Exercise Price in effect immediately prior to such subdivision will be proportionately reduced and the number of Warrant Shares will be proportionately increased. If the Company at any time on or after the Issuance Date combines (by combination, reverse stock split or otherwise) one or more classes of its outstanding shares of Common Stock into a smaller number of shares, the Exercise Price in effect immediately prior to such combination will be proportionately increased and the number of Warrant Shares will be proportionately decreased. Any adjustment under this Section 2 shall become effective at the close of business on the date the subdivision or combination becomes effective.
- 3. <u>RIGHTS UPON DISTRIBUTION OF ASSETS</u>. In addition to any adjustments pursuant to Section 2 above, if, on or after the Issuance Date and on or prior to the Expiration Date, the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property, options, evidence of indebtedness or any other assets by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations or restrictions on exercise of this Warrant, including without limitation, the Maximum Percentage) immediately before the date on which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution (provided, however, that to the extent that the Holder's right to participate in any such Distribution would result in the Holder and the other Attribution Parties exceeding the Maximum Percentage, then the Holder shall not be entitled to participate in such Distribution to such extent (and shall not be entitled to beneficial ownership) to such extent) and the portion of such Distribution Parties exceeding the Maximum Percentage, at which time or times as its right thereto would not result in the Holder and the other Attribution Parties exceeding the Maximum Percentage, at which time or times as its right thereto would not result in the Holder and the other Attribution Parties exce

4. PURCHASE RIGHTS; FUNDAMENTAL TRANSACTIONS.

(a) Purchase Rights. In addition to any adjustments pursuant to Section 2 above, if at any time on or after the Issuance Date and on or prior to the Expiration Date the Company grants, issues or sells any Options, Convertible Securities or rights to purchase stock, warrants, securities or other property pro rata to the record holders of any class of Common Stock (the "Purchase Rights"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations or restrictions on exercise of this Warrant, including without limitation, the Maximum Percentage) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of Common Stock are to be determined for the grant, issuance or sale of such Purchase Rights (provided, however, that to the extent that the Holder's right to participate in any such Purchase Right would result in the Holder and the other Attribution Parties exceeding the Maximum Percentage, then the Holder shall not be entitled to participate in such Purchase Right to such extent (and shall not be entitled to beneficial ownership) to such extent) and such Purchase Right to such extent shall be held in abeyance for the benefit of the Holder until such time or times as its right thereto would not result in the Holder and the other Attribution Parties exceeding the Maximum Percentage, at which time or times the Holder shall be granted such right (and any Purchase Right granted, issued or sold on such initial Purchase Right or on any subsequent Purchase Right to be held similarly in abeyance) to the same extent as if there had been no such limitation).

(b) Fundamental Transactions. The Company shall not enter into or be party to a Fundamental Transaction unless the Successor Entity assumes in writing, all of the obligations of the Company under this Warrant and all other Transaction Documents in accordance with the provisions of this Section 4(b), including agreements to deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant, but which is exercisable for a corresponding number of shares of capital stock equivalent to the shares of Common Stock issuable upon exercise of this Warrant (without regard to any limitations on the exercise of this Warrant) prior to such Fundamental Transaction, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (taking into account the relative value of the shares of Common Stock pursuant to such Fundamental Transaction and the value of such shares of capital stock, such adjustments to the number of shares of capital stock and such exercise price being for the purpose of protecting the economic value of this Warrant immediately prior to the consummation of such Fundamental Transaction). Upon the consummation of each Fundamental Transaction, the Successor Entity shall succeed to, and be substituted for the Company (so that from and after the date of such Fundamental Transaction, each and every provision of this Warrant referring to the "Company" shall instead refer to the Successor Entity), and the Successor Entity may exercise every prior right and power of the Company and shall assume all prior obligations of the Company under this Warrant with the same effect as if the Successor Entity had been named as the Company in this Warrant. On or prior to the consummation of each Fundamental Transaction, the Successor Entity shall deliver to the Holder confirmation that there shall be issued upon exercise of this Warrant at any time after the consummation of the Fundamental Transaction, in lieu of the shares of Common Stock (or other securities, cash, assets or other property purchasable upon the exercise of this Warrant prior to such Fundamental Transaction), such shares of stock, securities, cash,

assets or any other property whatsoever (including warrants or other purchase or subscription rights), which for purposes of clarification may continue to be shares of Common Stock, if any, that the Holder would have been entitled to receive upon the happening of such Fundamental Transaction or the record, eligibility or other determination date for the event resulting in such Fundamental Transaction, had this Warrant been exercised immediately prior to such Fundamental Transaction or the record, eligibility or other determination date for the event resulting in such Fundamental Transaction (without regard to any limitations on the exercise of this Warrant), as adjusted in accordance with the provisions of this Warrant. Notwithstanding the foregoing, and without limiting the provisions of Section 1(g) hereof, the Holder may elect, at its sole discretion, by delivery of a written notice to the Company, to permit a Fundamental Transaction without the required assumption of this Warrant. In addition to and not in substitution for any other rights hereunder, prior to the consummation of any Fundamental Transaction pursuant to which holders of Common Stock are entitled to receive securities, cash, assets or other property with respect to or in exchange for Common Stock (a "Corporate Event"), the Company shall make appropriate provision to ensure that the Holder will thereafter have the right to receive upon exercise of this Warrant at any time after the consummation of the Corporate Event, but prior to the Expiration Date, in lieu of the shares of Common Stock (or other securities, cash, assets or other property) (except such items still issuable under Sections 3 and 4(a), which shall continue to be receivable thereafter) issuable upon exercise of this Warrant prior to such Corporate Event, such shares of stock, securities, cash, assets or any other property whatsoever (including warrants or other purchase or subscription rights) which the Holder would have been entitled to receive upon the consummation of such Corporate Event or the record, eligibility or other determination date for the event resulting in such Corporate Event, had this Warrant been exercised immediately prior to such Corporate Event or the record, eligibility or other determination date for the event resulting in such Corporate Event (without regard to any limitations on exercise of this Warrant). Provision made pursuant to the preceding sentence shall be in a form and substance reasonably satisfactory to the Holder. The provisions of this Section 4(b) shall apply similarly and equally to successive Fundamental Transactions and Corporate Events.

(c) Notwithstanding the foregoing, in the event of Fundamental Transaction, at the request of the Holder delivered before the ninetieth (90th) day after the consummation of such Fundamental Transaction, the Company (or the Successor Entity) shall purchase this Warrant from the Holder by paying to the Holder, within five (5) Business Days after such request (or, if later, on the effective date of the Fundamental Transaction), cash in an amount equal to the Black Scholes Value of the remaining unexercised portion of this Warrant on the date of such Fundamental Transaction; provided, however, that if the Fundamental Transaction is not within the Company's control, including not approved by the Company's Board of Directors, the Holder shall only be entitled to receive from the Company or any Successor Entity, the same type or form of consideration (and in the same proportion), at the Black Scholes Value of the unexercised portion of this Warrant, that is being offered and paid to the holders of the Common Stock of the Company in connection with the Fundamental Transaction, whether that consideration be in the form of cash, securities or any combination thereof, or whether the holders of Common Stock are given the choice to receive from among alternative forms of consideration in connection with the Fundamental Transaction, such holders of Common Stock will be deemed to have received common stock of the Successor Entity (which entity may be the Company following such Fundamental Transaction) in such Fundamental Transaction.

- 5. NONCIRCUMVENTION. The Company hereby covenants and agrees that the Company will not, by amendment of its Amended and Restated Articles of Incorporation or Amended and Restated Bylaws, or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issuance or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, and will at all times in good faith carry out all of the provisions of this Warrant and take all action as may be required to protect the rights of the Holder. Without limiting the generality of the foregoing, the Company (i) shall not increase the par value of any shares of Common Stock receivable upon the exercise of this Warrant above the Exercise Price then in effect, (ii) shall take all such actions as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable shares of Common Stock upon the exercise of this Warrant, and (iii) shall, so long as any of the Warrants are outstanding, take all action necessary to reserve and keep available out of its authorized and unissued shares of Common Stock, solely for the purpose of effecting the exercise of the Warrants, the number of shares of Common Stock as shall from time to time be necessary to effect the exercise of the Warrants then outstanding (without regard to any limitations on exercise).
- 6. HOLDER NOT DEEMED A STOCKHOLDER. Except as otherwise specifically provided herein, the Holder, solely in such Person's capacity as a holder of this Warrant, shall not be entitled to vote or receive dividends or be deemed the holder of capital stock of the Company for any purpose, nor shall anything contained in this Warrant be construed to confer upon the Holder, solely in such Person's capacity as the Holder of this Warrant, any of the rights of a stockholder of the Company or any right to vote, give or withhold consent to any corporate action (whether any reorganization, issue of stock, reclassification of stock, consolidation, merger, conveyance or otherwise), receive notice of meetings, receive dividends or subscription rights, or otherwise, prior to the issuance to the Holder of the Warrant Shares which such Person is then entitled to receive upon the due exercise of this Warrant. In addition, nothing contained in this Warrant shall be construed as imposing any liabilities on the Holder to purchase any securities (upon exercise of this Warrant or otherwise) or as a stockholder of the Company, whether such liabilities are asserted by the Company or by creditors of the Company. Notwithstanding this Section 6, the Company shall provide the Holder with copies of the same notices and other information given to the stockholders of the Company generally, contemporaneously with the giving thereof to the stockholders.

7. REISSUANCE OF WARRANTS.

(a) <u>Transfer of Warrant</u>. If this Warrant is to be transferred, the Holder shall surrender this Warrant to the Company, whereupon the Company will forthwith issue and deliver upon the order of the Holder a new Warrant (in accordance with Section 7(d)), registered as the Holder may request, representing the right to purchase the number of Warrant Shares being transferred by the Holder and, if less than the total number of Warrant Shares then underlying this Warrant is being transferred, a new Warrant (in accordance with Section 7(d)) to the Holder representing the right to purchase the number of Warrant Shares not being transferred.

- (b) <u>Lost, Stolen or Mutilated Warrant</u>. Upon receipt by the Company of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant, and, in the case of loss, theft or destruction, of any indemnification undertaking by the Holder to the Company in customary form (but without the obligation to post a bond) and, in the case of mutilation, upon surrender and cancellation of this Warrant, the Company shall execute and deliver to the Holder a new Warrant (in accordance with Section 7(d)) representing the right to purchase the Warrant Shares then underlying this Warrant.
- (c) Exchangeable for Multiple Warrants. This Warrant is exchangeable, upon the surrender hereof by the Holder at the principal office of the Company, for a new Warrant or Warrants (in accordance with Section 7(d)) representing in the aggregate the right to purchase the number of Warrant Shares then underlying this Warrant, and each such new Warrant will represent the right to purchase such portion of such Warrant Shares as is designated by the Holder at the time of such surrender.
- (d) <u>Issuance of New Warrants</u>. Whenever the Company is required to issue a new Warrant pursuant to the terms of this Warrant, such new Warrant (i) shall be of like tenor with this Warrant, (ii) shall represent, as indicated on the face of such new Warrant, the right to purchase the Warrant Shares then underlying this Warrant (or in the case of a new Warrant being issued pursuant to Section 7(a) or Section 7(c), the Warrant Shares designated by the Holder which, when added to the number of shares of Common Stock underlying the other new Warrants issued in connection with such issuance, does not exceed the number of Warrant Shares then underlying this Warrant), (iii) shall have an issuance date, as indicated on the face of such new Warrant which is the same as the Issuance Date, and (iv) shall have the same rights and conditions as this Warrant.
- 8. NOTICES. Whenever notice is required to be given under this Warrant, including, without limitation, an Exercise Notice, unless otherwise provided herein, such notice shall be given in writing, (i) if delivered (a) from within the domestic United States, by first-class registered or certified airmail, or nationally recognized overnight express courier, postage prepaid, electronic mail or by facsimile or (b) from outside the United States, by International Federal Express, electronic mail or facsimile, and (ii) will be deemed given (A) if delivered by first-class registered or certified mail domestic, three (3) Business Days after so mailed, (B) if delivered by nationally recognized overnight carrier, one (1) Business Day after so mailed, (C) if delivered by International Federal Express, two (2) Business Days after so mailed and (D) if delivered by electronic mail, the time of transmission and (E) if delivered by facsimile, the time of transmission (provided that confirmation of transmission is generated and kept by the sending party), and will be delivered and addressed as follows:
 - (i) if to the Company, to:

Baudax Bio, Inc. 490 Lapp Road, Malvern, PA 19355 Attention: Gerri Henwood, Chief Executive Officer Email: ghenwood@baudaxbio.com¹

(ii) if to the Holder, at such address or other contact information delivered by the Holder to Company or as is on the books and records of the Company.

The Company shall provide the Holder with prompt written notice of all actions taken pursuant to this Warrant, including in reasonable detail a description of such action and the reason therefor.

Baudax to confirm notice details.

Without limiting the generality of the foregoing, the Company will give written notice to the Holder (i) immediately upon any adjustment of the Exercise Price, setting forth in reasonable detail, and certifying, the calculation of such adjustment and (ii) at least fifteen (15) days prior to (A) the date on which the Company closes its books or takes a record, with respect to any dividend or distribution upon the shares of Common Stock, with respect to any grants, issuances or sales of any Options, Convertible Securities or rights to purchase stock, warrants, securities or other property to holders of shares of Common Stock or for determining rights to vote with respect to any Fundamental Transaction, dissolution or liquidation; provided in each case that such information shall be made known to the public prior to or in conjunction with such notice being provided to the Holder, (B) any capital reorganization of the Company, any reclassification or recapitalization of the capital stock of the Company, any consolidation or merger involving the Company and any other Person or any transfer of all or substantially all the assets of the Company to any other Person, or any other Fundamental Transaction, or (C) any voluntary or involuntary dissolution, liquidation or winding-up of the Company, which notice will specify (1) the date or expected date on which any such record is to be taken for the purpose of such dividend, distribution or right, and the amount and character of such dividend, distribution or right, (2) the date or expected date on which any such reorganization, reclassification, recapitalization, consolidation, merger, transfer, or other Fundamental Transaction, or dissolution, liquidation or winding-up is to take place, and (3) the time, if any such time is to be fixed, as of which the holders of record of Common Stock shall be entitled to exchange their Common Stock for the securities or other property deliverable upon such reorganization, reclassification, recapitalization, consolidation, merger, transfer, or other Fundamental Transaction, or dissolution, liquidation or winding-up and a description in reasonable detail of the transaction. It is expressly understood and agreed that the time of exercise specified by the Holder in each Exercise Notice shall be definitive and may not be disputed or challenged by the Company.

- 9. <u>AMENDMENT AND WAIVER</u>. Except as otherwise provided herein, the provisions of this Warrant may be amended or waived and the Company may take any action herein prohibited, or omit to perform any act herein required to be performed by it, only if the Company has obtained the written consent of the Holder.
- 10. GOVERNING LAW; JURISDICTION; JURY TRIAL. This Warrant shall be governed by and construed and enforced in accordance with, and all questions concerning the construction, validity, interpretation and performance of this Warrant shall be governed by, the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of New York. The Company hereby irrevocably submits to the exclusive jurisdiction of the state and federal courts sitting in The City of New York, Borough of Manhattan, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. The Company hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to the Company at the address set forth in Section 8(i) above or such other address as the Company subsequently delivers to the Holder and agrees that such service shall constitute good and sufficient

service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Nothing contained herein shall be deemed or operate to preclude the Holder from bringing suit or taking other legal action against the Company in any other jurisdiction to collect on the Company's obligations to the Holder, to realize on any collateral or any other security for such obligations, or to enforce a judgment or other court ruling in favor of the Holder. If either party shall commence an action, suit or proceeding to enforce any provisions of this Warrant, the prevailing party in such action, suit or proceeding shall be reimbursed by the other party for their reasonable attorneys' fees and other costs and expenses incurred with the investigation, preparation and prosecution of such action or proceeding. THE COMPANY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS WARRANT OR ANY TRANSACTION CONTEMPLATED HEREBY.

- 11. <u>DISPUTE RESOLUTION</u>. In the case of a dispute as to the determination of the Exercise Price or the arithmetic calculation of the Warrant Shares, the Company shall submit the disputed determinations or arithmetic calculations via facsimile or electronic mail within two (2) Business Days of receipt of the Exercise Notice or other event giving rise to such dispute, as the case may be, to the Holder. If the Holder and the Company are unable to agree upon such determination or calculation of the Exercise Price or the Warrant Shares within three (3) Business Days of such disputed determination or arithmetic calculation being submitted to the Holder, then the Company shall, within two (2) Business Days submit via facsimile or electronic mail (a) the disputed determination of the Exercise Price to an independent, reputable investment bank selected by the Company and approved by the Holder or (b) the disputed arithmetic calculation of the Warrant Shares to the Company's independent, outside accountant. The Company shall cause at its expense the investment bank or the accountant, as the case may be, to perform the determinations or calculations and notify the Company and the Holder of the results no later than ten (10) Business Days from the time it receives the disputed determinations or calculations. Such investment bank's or accountant's determination or calculation, as the case may be, shall be binding upon all parties absent demonstrable error.
- 12. <u>REMEDIES, OTHER OBLIGATIONS, BREACHES AND INJUNCTIVE RELIEF.</u> The remedies provided in this Warrant shall be cumulative and in addition to all other remedies available under this Warrant and any other Transaction Documents, at law or in equity (including a decree of specific performance and/or other injunctive relief), and nothing herein shall limit the right of the Holder to pursue actual damages for any failure by the Company to comply with the terms of this Warrant. The Company acknowledges that a breach by it of its obligations hereunder will cause irreparable harm to the Holder and that the remedy at law for any such breach may be inadequate. The Company therefore agrees that, in the event of any such breach or threatened breach, the holder of this Warrant shall be entitled, in addition to all other available remedies, to an injunction restraining any breach, without the necessity of showing economic loss and without any bond or other security being required. Notwithstanding the foregoing or any else herein to the contrary, other than as expressly provided in Section 1(a), Section 1(c) or Section 2 hereof, if the Company is for any reason unable to issue and deliver Warrant Shares upon exercise of this Warrant as required pursuant to the terms hereof, the Company shall have no obligation to pay to the holder any cash or other consideration or otherwise "net cash settle" this Warrant; provided that the forgoing shall not limit or supersede the applicability of Section 4(b) hereof

- 13. <u>TRANSFER</u>. This Warrant and the Warrant Shares may be offered for sale, sold, transferred, pledged or assigned without the consent of the Company.
- 14. SEVERABILITY; CONSTRUCTION; HEADINGS. If any provision of this Warrant is prohibited by law or otherwise determined to be invalid or unenforceable by a court of competent jurisdiction, the provision that would otherwise be prohibited, invalid or unenforceable shall be deemed amended to apply to the broadest extent that it would be valid and enforceable, and the invalidity or unenforceability of such provision shall not affect the validity of the remaining provisions of this Warrant so long as this Warrant as so modified continues to express, without material change, the original intentions of the parties as to the subject matter hereof and the prohibited nature, invalidity or unenforceability of the provision(s) in question does not substantially impair the respective expectations or reciprocal obligations of the parties or the practical realization of the benefits that would otherwise be conferred upon the parties. The parties will endeavor in good faith negotiations to replace the prohibited, invalid or unenforceable provision(s) with a valid provision(s), the effect of which comes as close as possible to that of the prohibited, invalid or unenforceable provision(s). This Warrant shall be deemed to be jointly drafted by the Company and the Holder and shall not be construed against any Person as the drafter hereof. The headings of this Warrant are for convenience of reference and shall not form part of, or affect the interpretation of, this Warrant.
- 15. <u>DISCLOSURE</u>. Upon receipt or delivery by the Company of any notice in accordance with the terms of this Warrant, unless the Company has in good faith determined that the matters relating to such notice do not constitute material, nonpublic information relating to the Company or its subsidiaries, the Company shall contemporaneously with any such receipt or delivery publicly disclose such material, nonpublic information on a Current Report on Form 8-K or otherwise. In the event that the Company believes that a notice contains material, nonpublic information relating to the Company or its subsidiaries, the Company so shall indicate to such Holder contemporaneously with delivery of such notice, and in the absence of any such indication, the Holder shall be allowed to presume that all matters relating to such notice do not constitute material, nonpublic information relating to the Company or its subsidiaries.
 - 16. COVENANTS. The Company agrees that:
- (a) <u>Information</u>. So long as this Warrant remains outstanding or the Holder holds any Warrant Shares, the Company will deliver to the Holder:
 - (i) as soon as available and in any event within ninety (90) days after the end of each fiscal year, an audited consolidated balance sheet of the Company and its consolidated subsidiaries as of the end of such fiscal year and the related consolidated statements of income and cash flows for such fiscal year, prepared in conformity with generally accepted accounting principles in the United States applied on a consistent basis, and setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail and reported on without qualification by public accountants of nationally recognized standing;

- (ii) as soon as available but not later than forty-five (45) days after the end of the first three fiscal quarters of each fiscal year of the Company, a consolidated balance sheet of the Company as of the end of such quarter, and the related consolidated statements of income and cash flows for such quarter and for the portion of the fiscal year then ended, prepared in conformity with generally accepted accounting principles in the United States applied on a consistent basis, subject only to normal year-end audit adjustments and the absence of footnotes, and setting forth, in each case, in comparative form the figures for the corresponding quarter and the corresponding portion of the previous fiscal year;
- (iii) promptly upon the filing thereof, copies of all registration statements (other than the exhibits thereto and any registration statements on Form S-8 or its equivalent) and to the extent the Company is required by law or pursuant to the terms of any outstanding indebtedness of the Company to prepare such reports, any annual reports, quarterly reports and other periodic reports pursuant to Section 13 or 15(d) of the Exchange Act actually prepared by the Company as soon as available; and
 - (iv) promptly, upon the issuance thereof, all statements and notices sent to the Company's shareholders.

The information requirements set forth in Sections 16(a)(i)-(iv) shall be deemed to be satisfied upon filing of such information via EDGAR with the SEC.

- (b) Securities Filings; Rules 144 & 144A. The Company will (i) file any reports required to be filed by it under the Securities Act, the Exchange Act or the rules and regulations adopted by the SEC thereunder, (ii) use its commercially reasonable efforts to cooperate with the Holder and each holder of Warrant Shares in supplying such information concerning the Company as may be necessary for the Holder or holder of Warrant Shares to complete and file any information reporting forms currently or hereafter required by the SEC as a condition to the availability of an exemption from the Securities Act for the sale of any Warrants or Warrant Shares, (iii) take such further action as the Holder may reasonably request to the extent required from time to time to enable the Holder to sell Warrant Shares without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 or 144A under the Securities Act, as such rules may be amended from time to time, or any similar rule or regulation hereafter adopted by the SEC, and (iv) upon the request of the Holder, deliver to the Holder a written statement as to whether it has complied with such reporting requirements; provided that this subsection (b) shall not require the Company to make any filing under the Securities Act or Exchange Act which the Company is not otherwise obligated to make.
- (c) Obtaining of Governmental Approvals and Stock Exchange Listings. The Company will, at its own expense, (i) obtain and keep effective any and all permits, consents and approvals of governmental agencies and authorities which may from time to time be required of the Company in order to satisfy its obligations hereunder, and (ii) take all action which may be necessary so that the Warrant Shares, immediately upon their issuance upon the exercise of the Warrants, will be listed on the Company's primary trading market or quotation system with respect to the Common Stock, if any, on which the shares of Common Stock of the Company are then listed.

(d) [Reserved].

- (e) <u>Structural Dilution</u>. So long as this Warrant remains outstanding, the Company shall not permit any of its subsidiaries to issue, sell, distribute or otherwise grant in any manner (including by assumption) other than to any other subsidiary of the Company any rights to subscribe for or to purchase, or any warrants or options for the purchase of any equity securities of such subsidiary or any securities convertible into or exchangeable for such equity securities (or any rights to subscribe for or to purchase, or any warrants or options for the purchase of any such convertible or exchangeable securities), whether or not immediately exercisable or exercisable prior to the Expiration Date or thereafter.
- (f) Expenses. The Company shall pay all reasonable out-of-pocket expenses of the Holder, including reasonable fees and disbursements of counsel, in connection with the preparation of the Warrant, any waiver or consent hereunder or any amendment or modification hereof (regardless of whether the same becomes effective). Other than as set forth in the Credit Agreement, if applicable, the Company shall not be required to pay any expenses of the Holder under any other circumstance including those arising solely in connection with a transfer of the Warrant.
 - 17. CERTAIN DEFINITIONS. For purposes of this Warrant, the following terms shall have the following meanings:
- (a) "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by, or is under common control with, such Person, it being understood for purposes of this definition that "control" of a Person means the power directly or indirectly either to vote 10% or more of the stock having ordinary voting power for the election of directors of such Person or direct or cause the direction of the management and policies of such Person whether by contract or otherwise.
- (b) "Attribution Parties" means, collectively, the following Persons and entities: (i) any investment vehicle, including, any funds, feeder funds or managed accounts, currently, or from time to time after the Issuance Date, directly or indirectly managed or advised by the Holder's investment manager or any of its Affiliates or principals, (ii) any direct or indirect Affiliates of the Holder or any of the foregoing, (iii) any Person acting or who could be deemed to be acting as a Group together with the Holder or any of the foregoing and (iv) any other Persons whose beneficial ownership of the Company's Common Stock would or could be aggregated with the Holder's and the other Attribution Parties for purposes of Section 13(d) of the 1934 Act. For clarity, the purpose of the foregoing is to subject collectively the Holder and all other Attribution Parties to the Maximum Percentage.
- (c) "Bid Price" means, for any security as of the particular time of determination, the bid price for such security on the Principal Market as reported by Bloomberg as of such time of determination, or, if the Principal Market is not the principal securities exchange or trading market for such security, the bid price of such security on the principal securities exchange or trading market where such security is listed or traded as reported by Bloomberg as of such time of determination, or if the foregoing does not apply, the bid price of such security in the over-the-counter market on the electronic bulletin board for such security as reported by Bloomberg as of such time of determination, or, if no bid price is reported for such security by Bloomberg as of such time of determination, the average of the bid prices of any market makers

for such security as reported on the Pink Open Market as of such time of determination. If the Bid Price cannot be calculated for a security as of the particular time of determination on any of the foregoing bases, the Bid Price of such security as of such time of determination shall be the fair market value as mutually determined by the Company and the Holder. If the Company and the Holder are unable to agree upon the fair market value of such security, then such dispute shall be resolved in accordance with the procedures in Section 11 All such determinations shall be appropriately adjusted for any stock dividend, stock split, stock combination or other similar transaction during such period.

- (d) "Black Scholes Value" means the value of this Warrant based on the Black-Scholes Option Pricing Model obtained from the "OV" function on Bloomberg determined as of the day immediately following the first public announcement of the applicable Fundamental Transaction, or, if the Fundamental Transaction is not publicly announced, the date the Fundamental Transaction is consummated, for pricing purposes and reflecting (i) a risk-free interest rate corresponding to the U.S. Treasury rate for a period equal to the remaining term of this Warrant as of such date of request, (ii) an expected volatility equal to the greater of 100% and the 100 day volatility obtained from the HVT function on Bloomberg as of the day immediately following the public announcement of the applicable Fundamental Transaction, or, if the Fundamental Transaction is not publicly announced, the date the Fundamental Transaction is consummated, (iii) the underlying price per share used in such calculation shall be the greater of (i) the sum of the price per share being offered in cash, if any, plus the per share value of any non-cash consideration, if any, being offered in the Fundamental Transaction and (ii) the greater of (x) the last Weighted Average Price immediately prior to the public announcement of such Fundamental Transaction and (y) the last Weighted Average Price immediately prior to the consummation of such Fundamental Transaction, (iv) a zero cost of borrow and (v) a 360 day annualization factor.
 - (e) "Bloomberg" means Bloomberg Financial Markets.
- (f) "Business Day" means any day other than Saturday, Sunday or other day on which commercial banks in The City of New York are authorized or required by law to remain closed.
- (g) "Closing Bid Price" and "Closing Sale Price" means, for any security as of any date, the last closing bid price and last closing trade price, respectively, for such security on the Principal Market, as reported by Bloomberg, or, if the Principal Market begins to operate on an extended hours basis and does not designate the closing bid price or the closing trade price, as the case may be, then the last bid price or the last trade price, respectively, of such security prior to 4:00:00 p.m., New York time, as reported by Bloomberg, or, if the Principal Market is not the principal securities exchange or trading market for such security is listed or traded as reported by Bloomberg, or if the foregoing do not apply, the last closing bid price or last trade price, respectively, of such security in the over-the-counter market on the electronic bulletin board for such security as reported by Bloomberg, or, if no closing bid price or last trade price, respectively, is reported for such security by Bloomberg, the average of the bid prices, or the ask prices, respectively, of any market makers for such security as reported on the in the OTC Link or on the Pink Open Market. If the Closing Bid Price or the

Closing Sale Price cannot be calculated for a security on a particular date on any of the foregoing bases, the Closing Bid Price or the Closing Sale Price, as the case may be, of such security on such date shall be the fair market value as mutually determined by the Company and the Holder. If the Company and the Holder are unable to agree upon the fair market value of such security, then such dispute shall be resolved pursuant to Section 11. All such determinations to be appropriately adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction during the applicable calculation period.

- (h) "Common Stock" means (i) the Company's Common Stock, par value \$0.01 per share, and (ii) any capital stock into which such Common Stock shall have been changed or any capital stock resulting from a reclassification of such Common Stock.
- (i) "Convertible Securities" means any stock or securities (other than Options) directly or indirectly convertible into or exercisable or exchangeable for shares of Common Stock.
- (j) "Eligible Market" means The NASDAQ Capital Market, the NYSE American, The NASDAQ Global Select Market, The NASDAQ Global Market or The New York Stock Exchange, Inc.
- (k) "Exchange Act" means the Securities Exchange Act of 1934, or any successor federal statute, and the rules and regulations of the SEC thereunder, all as the same shall be in effect at the time.
- (1) "Expiration Date" means the date ten (10) years after the Issuance Date or, if such date falls on a day other than a Business Day or on which trading does not take place on the Principal Market (a "Holiday"), the next day that is not a Holiday.
- (m) "Fundamental Transaction" means (A) that the Company shall, directly or indirectly, including through subsidiaries, Affiliates or otherwise, in one or more related transactions, (i) consolidate or merge with or into (whether or not the Company is the surviving corporation) another Subject Entity, or (ii) sell, assign, transfer, convey or otherwise dispose of all or substantially all of the properties or assets of the Company or any of its "significant subsidiaries" (as defined in Rule 1-02 of Regulation S-X) to one or more Subject Entities, or (iii) make, or allow one or more Subject Entities to make, or allow the Company to be subject to or have its shares of Common Stock be subject to or party to one or more Subject Entities making, a purchase, tender or exchange offer that is accepted by the holders of at least either (x) 50% of the outstanding shares of Common Stock, (y) 50% of the outstanding shares of Common Stock calculated as if any shares of Common Stock held by all Subject Entities making or party to, or Affiliated with any Subject Entities making or party to, such purchase, tender or exchange offer were not outstanding; or (z) such number of shares of Common Stock such that all Subject Entities making or party to, or Affiliated with any Subject Entity making or party to, such purchase, tender or exchange offer, become collectively the beneficial owners (as defined in Rule 13d-3 under the 1934 Act) of at least 50% of the outstanding shares of Common Stock, or (iv) consummate a stock purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off or scheme of arrangement) with one or more Subject Entities whereby all such Subject Entities, individually or in the aggregate, acquire, either (x) at least 50% of the

outstanding shares of Common Stock, (y) at least 50% of the outstanding shares of Common Stock calculated as if any shares of Common Stock held by all the Subject Entities making or party to, or Affiliated with any Subject Entity making or party to, such stock purchase agreement or other business combination were not outstanding; or (z) such number of shares of Common Stock such that the Subject Entities become collectively the beneficial owners (as defined in Rule 13d-3 under the 1934 Act) of at least 50% of the outstanding shares of Common Stock, or (v) reorganize, recapitalize or reclassify its shares of Common Stock (other than a reorganization, recapitalization or reclassification subject to Section 2), (B) that the Company shall, directly or indirectly, including through subsidiaries, Affiliates or otherwise, in one or more related transactions, allow any Subject Entity individually or the Subject Entities in the aggregate to be or become the "beneficial owner" (as defined in Rule 13d-3 under the 1934 Act), directly or indirectly, whether through acquisition, purchase, assignment, conveyance, tender, tender offer, exchange, reduction in outstanding shares of Common Stock, merger, consolidation, business combination, reorganization, recapitalization, spin-off, scheme of arrangement, reorganization, recapitalization or reclassification or otherwise in any manner whatsoever, of either (x) at least 50% of the aggregate ordinary voting power represented by issued and outstanding shares of Common Stock, (y) at least 50% of the aggregate ordinary voting power represented by issued and outstanding shares of Common Stock not held by all such Subject Entities as of the Issuance Date calculated as if any shares of Common Stock held by all such Subject Entities were not outstanding, or (z) a percentage of the aggregate ordinary voting power represented by issued and outstanding shares of Common Stock or other equity securities of the Company sufficient to allow such Subject Entities to effect a statutory short form merger or other transaction requiring other stockholders of the Company to surrender their Common Stock without approval of the stockholders of the Company or (C) directly or indirectly, including through subsidiaries, Affiliates or otherwise, in one or more related transactions, the issuance of or the entering into any other instrument or transaction structured in a manner to circumvent, or that circumvents, the intent of this definition in which case this definition shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this definition to the extent necessary to correct this definition or any portion of this definition which may be defective or inconsistent with the intended treatment of such instrument or transaction.

- (n) "Group" means a "group" as that term is used in Section 13(d) of the 1934 Act and as defined inRule 13d-5 thereunder.
- (o) "Options" means any rights, warrants or options to subscribe for or purchase shares of Common Stock or Convertible Securities.
- (p) "Parent Entity" of a Person means an entity that, directly or indirectly, controls the applicable Person, including such entity whose common stock or equivalent equity security is quoted or listed on an Eligible Market (or, if so elected by the Holder, any other market, exchange or quotation system), or, if there is more than one such Person or such entity, the Person or such entity designated by the Holder or in the absence of such designation, such Person or entity with the largest public market capitalization as of the date of consummation of the Fundamental Transaction.

- (q) "Person" means an individual, a limited liability company, a partnership, a joint venture, a corporation, a trust, an unincorporated organization, any other entity and a government or any department or agency thereof.
 - (r) "Principal Market" means The NASDAQ Capital Market.
- (s) "Securities Act" means the Securities Act of 1933, or any successor federal statute, and the rules and regulations of the SEC thereunder, all as the same shall be in effect at the time.
- (t) "Standard Settlement Period" means the standard settlement period, expressed in a number of Trading Days, for the Company's primary trading market or quotation system with respect to the Common Stock that is in effect on the date of delivery of an applicable Exercise Notice.
 - (u) "Subject Entity" means any Person, Persons or Group or any Affiliate or associate of any such Person, Persons or Group.
- (v) "Successor Entity" means one or more Person or Persons (or, if so elected by the Holder, the Company or Parent Entity) formed by, resulting from or surviving any Fundamental Transaction or one or more Person or Persons (or, if so elected by the Holder, the Company or the Parent Entity) with which such Fundamental Transaction shall have been entered into.
- (w) "Trading Day" means any day on which the Common Stock is traded on the Principal Market, or, if the Principal Market is not the principal trading market for the Common Stock, then on the principal securities exchange or securities market on which the Common Stock is then traded.
 - (x) "Transaction Documents" means any agreement entered into by and between the Company and the Holder, as applicable.
- (y) "Weighted Average Price" means, for any security as of any date, the dollar volume-weighted average price for such security on the Principal Market during the period beginning at 9:30:00 a.m., New York time (or such other time as the Principal Market publicly announces is the official open of trading), and ending at 4:00:00 p.m., New York time (or such other time as the Principal Market publicly announces is the official close of trading), as reported by Bloomberg through its "Volume at Price" function or, if the foregoing does not apply, the dollar volume-weighted average price of such security in the over-the-counter market on the electronic bulletin board for such security during the period beginning at 9:30:00 a.m., New York time (or such other time as such market publicly announces is the official open of trading), and ending at 4:00:00 p.m., New York time (or such other time as such market publicly announces is the official close of trading), as reported by Bloomberg, or, if no dollar volume-weighted average price is reported for such security by Bloomberg for such hours, the average of the highest Closing Bid Price and the lowest closing ask price of any of the market makers for such security as reported in the OTC Link or on the Pink Open Market. If the Weighted Average Price cannot be calculated for a security on a particular date on any of the foregoing bases, the Weighted Average Price of such security on such date shall be the fair market value as mutually determined by the Company

and the Holder. If the Company and the Holder are unable to agree upon the fair market value of such security, then such dispute shall be resolved pursuant to Section 11 with the term "Weighted Average Price" being substituted for the term "Exercise Price." All such determinations shall be appropriately adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction during the applicable calculation period.

[Signature Page Follows]

IN WITNESS WHEREOF, the Company has caused this Warrant to Purchase Common Stock to be duly executed as of the Issuance Date set out above.

BAUDAX BIO, INC.

By: /s/ Gerri Henwood Name: Gerri Henwood

Title: Chief Executive Officer

EXERCISE NOTICE

TO BE EXECUTED BY THE REGISTERED HOLDER TO EXERCISE THIS WARRANT TO PURCHASE COMMON STOCK

BAUDAX BIO, INC.

The undersigned holder hereby exercises the right to purchase of the shares of Common Stock (Warrant Shares ") of Baudax Bio, Inc., a company organized under the laws of Pennsylvania (the " Company "), evidenced by the attached Warrant to Purchase Common Stock (the									
"Warrant"). Capitalized terms used herein and not otherwise defined shall have the respective meanings set forth in the Warrant.									
1. Form of Exercise Price. The Holder intends that payment of the Exercise Price shall be made as:									
	a "Cash Exercise" with respect to	Warrant Shares; and/or							
	a "Cashless Exercise" with respect to	Warrant Shares.							
2. Payment of Exercise Price. In the event that the holder has elected a Cash Exercise with respect to some or all of the Warrant Shares to be issued pursuant hereto, the holder shall pay the Aggregate Exercise Price in the sum of \$ to the Company in accordance with the terms of the Warrant.									
3. Delivery of Warrant Shares. T	the Company shall deliver to the holder	Warrant Shares in accordance with the terms of the Warrant.							
Date:,,									
Name of Registered Holder									
By:									
Title:									

ACKNOWLEDGMENT

The Company hereby acknowledges this Exercise Notice and hereby directs Broadridge Corporate Issuer Solutions, Inc. to issue the above indicated number of shares of Common Stock on or prior to the applicable Share Delivery Date.

BAUDAX BIO, INC.						
By:						
Name:						
Title:						

AMENDMENT NO. 5 AND CONSENT TO CREDIT AGREEMENT

This Amendment No. 5 and Consent to Credit Agreement (this "Amendment") dated as of March 29, 2023, is among Baudax Bio, Inc., a Pennsylvania corporation ("Borrower"), Baudax Bio N.A. LLC, a Delaware limited liability company ("Baudax LLC"), Baudax Bio Limited, a private company incorporated under the laws of Ireland limited by shares having company number 562027 (together with Baudax LLC, collectively, the "Guarantors" and together with the Borrower, the "Loan Parties"), Wilmington Trust, National Association, not individually, but solely in its capacity as administrative and collateral agent for the Lenders (the "Agent") and the Lenders party hereto (constituting all of the Lenders under the Credit Agreement (as defined below)).

WHEREAS, the Borrower, the Lenders and the Agent are party to that certain Credit Agreement, dated as of May 29, 2020, as amended by that certain Amendment No. 1 and Waiver to Credit Agreement, dated as of August 1, 2022, that certain Amendment No. 2 to Credit Agreement, dated as of October 24, 2022, that certain Amendment No. 3 to Credit Agreement, dated as of November 30, 2022 and the certain Amendment No. 4 to the Credit Agreement, dated as of January 5, 2023 (as it may be further amended, restated, amended and restated, supplemented or otherwise modified from time to time, collectively, the "Credit Agreement"), pursuant to which the Lenders agreed to make loans to the Borrower on the terms set forth therein;

WHEREAS, the Borrower desires to enter into the Permitted Alkermes Asset Transfer Transaction (as defined below) and the Agent and the Lenders have agreed to consent to the Permitted Alkermes Asset Transfer Transaction and make certain amendments to the Credit Agreement, on the terms and subject to the conditions set forth herein.

NOW, **THEREFORE**, in consideration of the mutual agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. <u>Definitions; Loan Document</u>. Capitalized terms used herein without definition shall have the meanings assigned to such terms in the Credit Agreement. This Amendment shall constitute a Loan Document for all purposes of the Credit Agreement and the other Loan Documents.
- 2. Consent. Notwithstanding the terms of Sections 7.4(b), 7.5(b), 7.5(c) and 8.1 of the Credit Agreement, the Agent and the Lenders hereby consent to the Permitted Alkermes Asset Transfer Transaction. The above consent shall not modify or affect the Loan Parties' obligations to comply fully with the terms of Sections 7.4(b), 7.5(b), 7.5(c) and 8.1 of the Credit Agreement or any other duty, term, condition or covenant contained in the Credit Agreement or any other Loan Document in the future. The consent is limited solely to the specific consent identified above and nothing contained in this Agreement shall be deemed to constitute a waiver of any other rights or remedies the Agent or any Lender may have under the Credit Agreement or any other Loan Document or under applicable law. "Permitted Alkermes Asset Transfer Transaction" means (a) the transfer of certain assets (including inventory and equipment) related the Borrower's product, Anjeso, pursuant to that certain Asset Transfer Agreement by and between Borrower and APIL (together with all exhibits and schedules thereto), dated as of the date hereof (the "Asset Transfer Agreement"), (b) the related termination of the Meloxicam Acquisition Agreement, the Meloxicam Transfer Agreement and the Development Agreement, (c) the related assignment to APIL of the Manufacturing Agreement and the Product Agreement and (d) all other transactions contemplated by the Asset Transfer Agreement.

- 3. Release of Liens. Upon the occurrence of the Fifth Amendment Effective Date, any security interest or other Liens granted to or held by the Agent under any Loan Document in respect of the Acquired Assets and Acquired Regulatory Approvals (as such terms are defined in the Asset Transfer Agreement) shall immediately be automatically, permanently and irrevocably released and discharged (such released assets, the "Released Assets") (collectively, the "Release"). The Agent agrees to execute and deliver to the Borrower, at the expense of the Borrower, any and all documents or instruments reasonably requested by the Borrower to further evidence or effectuate the Release. Upon the occurrence of the Release, the Agent hereby authorizes the Borrower (or any designee of the Borrower, including, without limitation, APIL and the attorneys or other agents of the Borrower or APIL) to file the UCC-3 amendment attached hereto as Exhibit A, the intellectual property release attached hereto as Exhibit B and other instruments, releases and documents evidencing the release of the Agent's security interests and other Liens in the Released Assets. The release set forth in this Section 3 is applicable only to the Released Assets and shall not affect, modify or diminish any obligations of any other Loan Party under the Credit Agreement and (B) shall not be deemed to be a modification to the Loan Documents except as set forth herein.
 - 4. Amendments. Upon the effectiveness of this Amendment, the Credit Agreement is hereby amended as follows:
 - (a) Section 1.1 is amended by replacing the definition of "Key Agreements" with the below:
 - ""Key Agreements" means any agreement entered into by the Loan Parties the cancellation of which would reasonably be expected to result in a Material Adverse Effect."
 - (b) The definition of "Obligations" in Section 1.1 is amended by inserting "Fifth Amendment Fee," immediately after the phrase "Facility

Fee,".

- (c) Section 1.1 is amended by replacing the definition of "Warrants" with the below:
- ""Warrants" means each of (i) the Closing Warrants, and (ii) the Fifth Amendment Warrants."
- (d) Section 1.1 is amended by adding the following definition in alphabetical order:
- ""Closing Warrants" means those certain 7-year common stock purchase warrants issued by the Borrower to the Lenders or Lenders' Affiliates on the Closing Date in accordance with Section 2.12.3."
- ""Fifth Amendment Effective Date" means March 29, 2023."
- ""Fifth Amendment Fee" has the meaning set forth in Section 2.12.6."
- ""Fifth Amendment Warrants" means those certain 10-year common stock purchase warrants with an exercise price of \$1.8951 issued by the Borrower to the Lenders or Lenders' Affiliates on the Fifth Amendment Effective Date."
- ""Required Lenders" means, at any time, Lenders having Loans representing more than 50% of the aggregate principal amount of Loans of all Lenders at such time."
- (e) Section 2.3.4(b) is hereby amended by replacing each reference therein to "Warrants" with "Closing Warrants".

- (f) Section 2.12 is amended by adding a new Section 2.12.6 as follows:
- "2.12.6 <u>Fifth Amendment Fee</u>. The Borrower shall pay to the Agent for the ratable distribution to the Lenders an amendment fee (the <u>Fifth Amendment Fee</u>") in the aggregate amount of \$567,500, which Fifth Amendment Fee shall be fully earned on the Fifth Amendment Effective Date, and due and payable in equal installments on the first Business Day of each of the 12 consecutive months immediately following the Fifth Amendment Effective Date."
- (g) Section 7.17.1 is amended and restated in its entirety as follows:
- "7.17.1 <u>Liquidity Accounts</u>. Not suffer or permit the aggregate amount of cash in the Liquidity Accounts to be less than \$2,500,000 at any time."
- (h) Section 8.11.14 is amended by and restated as follows:
- "RESERVED"
- (i) Attached hereto as Annex A is an updated Schedule 1.1(a). Schedule 1.1(a) is hereby amended to include the information set forth on Annex A hereto.
- 5. <u>Consideration</u>. As consideration for the agreements of the Agent and the Lenders herein, the Borrower agrees to (a) issue and deliver to the Lenders, for their own account on the Fifth Amendment Effective Date, the Fifth Amendment Warrants, and (b) pay to the Lenders the Fifth Amendment Fee in accordance with the terms of the Credit Agreement and this Amendment.
 - 6. Conditions to Effectiveness. This Amendment shall become effective on the date (such date, the 'Fifth Amendment Effective Date') on which:
 - the Agent and the Lenders receive counterpart signatures to this Amendment duly executed and delivered by the Loan Parties, the Agent and each Lender;
 - (b) the Agent and the Lenders receive a fully executed copy of the Asset Transfer Agreement and all documents entered into in connection therewith;
 - the Agent and the Lenders receive counterpart signatures to the Fifth Amendment Warrants duly executed and delivered by the Borrower;
 - (d) the Agent and the Lenders receive counterpart signatures to that certain Registration Rights Agreement dated as of the date hereof and duly executed and delivered by the Borrower and MAM Eagle Lender, LLC;
 - (e) substantially simultaneously the closing and consummation of the transactions contemplated by the Asset Transfer Agreement occurs;
 - (f) the Agent and the Lenders receive (i) a certificate of the Borrower, dated as of the Fifth Amendment Effective Date and executed by a secretary, assistant secretary, president, vice president, chief executive officer, chief financial officer or other duly appointed officers thereof, which shall (A) certify the full force and validity of the Borrower's articles of incorporation and bylaws and attach copies thereof, and (B) certify that attached thereto is a true and complete copy of resolutions or

- written consents of its board of directors authorizing the execution, delivery and performance of this Amendment and the Fifth Amendment Warrant, and that such resolutions or written consents have not been modified, rescinded or amended and are in full force and effect without amendment, modification or rescission, and (ii) a good standing certificate as of a recent date for such Credit Party from the Secretary of State for the Commonwealth of Pennsylvania;
- (g) Agent and the Lenders receive payment for all fees and reasonable and documented out-of-pocket expenses (to the extent Borrowers receive an invoice at least one (1) Business Day prior to the Fifth Amendment Effective Date) incurred in connection with the negotiation, preparation, execution and delivery of this Amendment and the Fifth Amendment Warrant and the registration rights agreement entered into in connection therewith; and
- (h) the representations and warranties in Section 7 shall be true and correct.
- 7. Representations and Warranties. The Loan Parties represent and warrant to the Lenders and the Agent that, after giving effect to this Amendment:
- (a) The representations and warranties of the Loan Parties contained in the Credit Agreement or any other Loan Document are true, accurate and correct in all material respects (without duplication of any materiality qualifiers); provided, however, that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects (without duplication of any materiality qualifiers) as of such date.
- (b) No Default or Event of Default under the Loan Documents has occurred and is continuing or would result from the effectiveness of this Amendment.
- 8. No Implied Amendment or Waiver. Except as expressly set forth in this Amendment, this Amendment is limited to the matters specifically set forth herein and shall not, by implication or otherwise, limit, impair, constitute a waiver of or otherwise affect any rights or remedies of the Agent or any Lender under the Loan Documents, or alter, modify, amend or in any way affect any of the terms, obligations or covenants contained in the Loan Documents, all of which shall continue in full force and effect. Nothing in this Amendment shall be construed to imply any willingness on the part of the Agent or any Lender to agree to or grant any similar or future amendment, consent or waiver of any of the terms and conditions of the Loan Document.
- 9. Waiver and Release. TO INDUCE THE AGENT AND THE LENDERS TO AGREE TO THE TERMS OF THIS AMENDMENT, EACH LOAN PARTY AND ITS AFFILIATES (COLLECTIVELY, THE "RELEASING PARTIES") REPRESENT AND WARRANT THAT, AS OF THE DATE HEREOF, THERE ARE NO CLAIMS OR OFFSETS AGAINST, OR RIGHTS OF RECOUPMENT WITH RESPECT TO, OR DISPUTES OF, OR DEFENSES OR COUNTERCLAIMS TO, THEIR OBLIGATIONS UNDER THE LOAN DOCUMENTS, AND IN ACCORDANCE THEREWITH THEY.
- (a) WAIVE ANY AND ALL SUCH CLAIMS, OFFSETS, RIGHTS OF RECOUPMENT, DISPUTES, DEFENSES AND COUNTERCLAIMS, WHETHER KNOWN OR UNKNOWN, ARISING PRIOR TO THE DATE HEREOF.

- (b) FOREVER RELEASE, RELIEVE AND DISCHARGE THE AGENT, EACH LENDER AND THEIR RESPECTIVE OFFICERS, DIRECTORS, SHAREHOLDERS, MEMBERS, PARTNERS, PREDECESSORS, SUCCESSORS, ASSIGNS, ATTORNEYS, ACCOUNTANTS, AGENTS, EMPLOYEES AND REPRESENTATIVES (COLLECTIVELY, THE "RELEASED PARTIES"), AND EACH OF THEM, FROM ANY AND ALL CLAIMS, LIABILITIES, DEMANDS, CAUSES OF ACTION, DEBTS, OBLIGATIONS, PROMISES, ACTS, AGREEMENTS AND DAMAGES, OF WHATEVER KIND OR NATURE, WHETHER KNOWN OR UNKNOWN, SUSPECTED OR UNSUSPECTED, CONTINGENT OR FIXED, LIQUIDATED OR UNLIQUIDATED, MATURED OR UNMATURED, WHETHER AT LAW OR IN EQUITY, WHICH THE RELEASING PARTIES EVER HAD, NOW HAVE, OR MAY, SHALL OR CAN HEREAFTER HAVE, DIRECTLY OR INDIRECTLY ARISING OUT OF OR IN ANY WAY BASED UPON, CONNECTED WITH, OR RELATED TO MATTERS, THINGS, ACTS, CONDUCT AND/OR OMISSIONS AT ANY TIME FROM THE BEGINNING OF THE WORLD THROUGH AND INCLUDING THE DATE HEREOF, INCLUDING WITHOUT LIMITATION ANY AND ALL CLAIMS AGAINST THE RELEASED PARTIES ARISING UNDER OR RELATED TO ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREBY.
- (c) IN CONNECTION WITH THE RELEASE CONTAINED HEREIN, ACKNOWLEDGE THAT THEY ARE AWARE THAT THEY MAY HEREAFTER DISCOVER CLAIMS PRESENTLY UNKNOWN OR UNSUSPECTED, OR FACTS IN ADDITION TO OR DIFFERENT FROM THOSE WHICH THEY KNOW OR BELIEVE TO BE TRUE, WITH RESPECT TO THE MATTERS RELEASED HEREIN. NEVERTHELESS, IT IS THE INTENTION OF THE RELEASING PARTIES, THROUGH THIS AMENDMENT AND WITH ADVICE OF COUNSEL, FULLY, FINALLY AND FOREVER TO RELEASE ALL SUCH MATTERS, AND ALL CLAIMS RELATED THERETO, WHICH DO NOW EXIST, OR HERETOFORE HAVE EXISTED. IN FURTHERANCE OF SUCH INTENTION, THE RELEASES HEREIN GIVEN SHALL BE AND REMAIN IN EFFECT AS A FULL AND COMPLETE RELEASE OF SUCH MATTERS NOTWITHSTANDING THE DISCOVERY OR EXISTENCE OF ANY SUCH ADDITIONAL OR DIFFERENT CLAIMS OR FACTS RELATED THERETO.
- (d) COVENANT AND AGREE NOT TO BRING ANY CLAIM, ACTION, SUIT OR PROCEEDING AGAINST THE RELEASED PARTIES, DIRECTLY OR INDIRECTLY, REGARDING OR RELATED IN ANY MANNER TO THE MATTERS RELEASED HEREBY, AND FURTHER COVENANT AND AGREE THAT THIS AGREEMENT IS A BAR TO ANY SUCH CLAIM, ACTION, SUIT OR PROCEEDING.
- (e) REPRESENT AND WARRANT TO THE RELEASED PARTIES THAT THEY HAVE NOT HERETOFORE ASSIGNED OR TRANSFERRED, OR PURPORTED TO ASSIGN OR TRANSFER, TO ANY PERSON OR ENTITY ANY CLAIMS OR OTHER MATTERS HEREIN RELEASED.
- (f) ACKNOWLEDGE THAT THEY HAVE HAD THE BENEFIT OF INDEPENDENT LEGAL ADVICE WITH RESPECT TO THE ADVISABILITY OF ENTERING INTO THIS RELEASE AND HEREBY KNOWINGLY, AND UPON SUCH ADVICE OF COUNSEL, WAIVE ANY AND ALL APPLICABLE RIGHTS AND BENEFITS UNDER, AND PROTECTIONS OF, CALIFORNIA CIVIL CODE SECTION 1542, AND ANY AND ALL STATUTES AND DOCTRINES OF SIMILAR EFFECT. CALIFORNIA CIVIL CODE SECTION 1542 PROVIDES AS FOLLOWS:

A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release, and that if known by him or her, would have materially affected his or her settlement with the debtor or released party.

- 10. Expenses. The Loan Parties agree to pay all reasonable and documented out-of-pocket costs and expenses of the Agent and the Lenders (including diligence costs, consulting fees and Costs) in connection with the transactions contemplated by this Agreement invoiced to the Borrower (including the reasonable and documented out-of-pocket fees and expenses of counsel to the Agent and the Lenders incurred in connection with the negotiation, preparation, execution and delivery of this Amendment and the other Loan Documents).
- 11. <u>Guarantor Reaffirmation</u>. Each Guarantor hereby ratifies and reaffirms as of the date hereof the guarantee granted by it to the Agent for the benefit of the Lenders under the Loan Documents and agrees and acknowledges that such guarantee shall continue and shall remain in full force and effect from and after the date hereof after giving effect from and after the date hereof, and the obligations guaranteed thereby shall include the Loan Parties' obligations under the Loan Documents from and after the date hereof. Except as expressly provided herein, this Amendment shall not release, reduce or diminish any Loan Party's obligations to the Agent and the Lenders under the Loan Documents, or prejudice, alter or in any regard adversely affect the rights and remedies of the Agent or any Lender in respect thereof.
- 12. Reaffirmation of Security Interest. Except as expressly set forth in this Amendment (including, with respect to the Release), each Loan Party hereby (i) affirms that each of the security interests and liens granted in or pursuant to the Loan Documents are valid and subsisting and shall continue and shall remain in full force and effect from and after the date hereof and (ii) agree that this Agreement shall in no manner impair or otherwise adversely affect any of the security interests and liens granted in or pursuant to the Loan Documents.
- 13. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one agreement. Executed copies of the signature pages of this Amendment sent by facsimile or transmitted electronically shall be treated as originals, fully binding and with full legal force and effect, and the parties waive any rights they may have to object to such treatment.
- 14. <u>Governing Law.</u> THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTION 5-1401 OF THE NEW YORK GENERAL OBLIGATIONS LAW).
- 15. Agent Authorization. Each of the undersigned Lenders, who collectively constitute all of the Lenders under the Credit Agreement, hereby (i) authorizes and directs the Agent to execute and deliver (A) this Amendment and (B) the intellectual property release attached hereto as Exhibit B, (ii) authorizes and directs the Agent to deliver the UCC-3 amendment attached hereto as Exhibit A, (iii) authorizes and directs the Agent to execute and/or deliver any other instruments, releases and documents evidencing the release of the Agent's security interests and other Liens in the Released Assets, (iv) acknowledges and agrees that the undersigned Lenders constitute all of the Lenders necessary to direct the Agent to execute such documents; and (v) acknowledges and agrees that the direction set forth in this Amendment constitutes an instruction, consent and request of the Lenders under the Loan Documents, including Sections 9.3 and 9.7 of the Credit Agreement.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the date first written above.

BORROWER:

BAUDAX BIO, INC.

By /s/ Gerri Henwood

Name: Gerri Henwood Title: Chief Executive Officer

GUARANTORS:

BAUDAX BIO N.A. LLC

By /s/ Gerri Henwood

Name: Gerri Henwood Title: Chief Executive Officer

BAUDAX BIO LIMITED

By /s/ Gerri Henwood

Name: Gerri Henwood Title: Chief Executive Officer

[Signature Page to Amendment No. 5 and Consent to Credit Agreement]

AGENT:

WILMINGTON TRUST, NATIONAL ASSOCIATION

By /s/ Andrew Lennon Name: Andrew Lennon Title: Assistance Vice President

LENDERS:

MAM EAGLE LENDER, LLC

By /s/ Lou Hanover

Name: Lou Hanover Title: Authorized Signatory

[Signature Page to Amendment No. 5 and Consent to Credit Agreement]

EXHIBIT A

UCC-3 Amendment

See attached.

EXHIBIT B

Partial Release of Intellectual Property Security Agreement

See attached.

ANNEX A

SCHEDULE 1.1(a)

Commitment Schedule

Commitments

	Tranche One	Tranche Two	Tranche Three	Tranche Four	Tranche Five
Lender	Commitments	Commitments	Commitments	Commitments	Commitments
MAM Eagle Lender, LLC	\$10,000,000	\$ 5,000,000	\$ 5,000,000	\$10,000,000	\$20,000,000
Total:	\$10,000,000	\$ 5,000,000	\$ 5,000,000	\$10,000,000	\$20,000,000

Closing Warrants¹

Lender	Warrants
MAM Eagle Lender, LLC	527,100
Total:	527,100

Fifth Amendment Warrants

Lender	Warrants
MAM Eagle Lender, LLC	785,026
Total:	785,026

Reflects the number of Closing Warrants issued on the Closing Date.

REGISTRATION RIGHTS AGREEMENT

This REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of March 29, 2023, is entered into by and between Baudax Bio, Inc., a company organized under the laws of the Commonwealth of Pennsylvania (the "Company"), and MAM Eagle Lender, LLC (the "Initial Holder").

RECITALS

WHEREAS, on or about the date hereof, the Company issued to the Initial Holder that certain Warrant to Purchase Common Stock, pursuant to which, among other things, the Initial Holder is entitled, subject to the terms and conditions set forth therein, to purchase from the Company shares of common stock, \$0.01 par value per share, of the Company (the "Company Common Stock"); and

WHEREAS, the Company has agreed to provide the Initial Holder with the registration rights specified in this Agreement with respect to Registrable Securities (as defined herein), on the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter contained and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1 DEFINITIONS

- 1.1 <u>Definitions</u>. The following terms shall have the meanings set forth in this <u>Section 1.1</u>:
- "Additional Warrant Securities" means any additional warrants issued to any Holder in accordance with Section 7 of the Warrant.
- "Adverse Effect" has the meaning given such term in Section 2.1.5 herein.
- "Advice" has the meaning given such term in Section 2.6 herein.

"Affiliate" means with respect to a party hereto, any Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such party. For purposes of this definition, "control" and, with correlative meanings, the terms "controlled by" and "under common control with" as used with respect to a Person means (a) the possession, directly or indirectly, of the power to direct, or cause the direction of, the management or policies of such Person, whether through the ownership of voting securities, by contract relating to voting rights or corporate governance, or otherwise, or (b) the ownership, directly or indirectly, of more than 50% of the voting securities or other ownership interest of a Person.

"Agreement" has the meaning given such term in the introductory paragraph of this Agreement.

- "Block Sale" means the sale of shares of Company Common Stock to one of several purchasers in a registered transaction by means of a bought deal, a block trade or a direct sale.
- "Business Days" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by law to be closed in New York City.
- "Company" has the meaning given such term in the introductory paragraph of this Agreement and includes the Company's successors by merger, acquisition, reorganization or otherwise.
 - "Company Common Stock" has the meaning given such term in the recitals of this Agreement.
 - "Company Indemnified Person" has the meaning given such term in Section 2.8.2 herein.
 - "Demand Registration" has the meaning given such term in Section 2.2.1(a) herein.
 - "Demand Request" has the meaning given such term in Section 2.2.1(a) herein.
 - "Demanding Shareholders" has the meaning given such term in Section 2.2.1(a) herein.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended, or any similar federal statute, and the rules and regulations promulgated by the SEC thereunder.
- "Excluded Registration" means a registration under the Securities Act (i) of Registrable Securities pursuant to one or more Demand Registrations pursuant to Section 2 hereof, (ii) of equity securities issuable in connection with the Company's stock option or other employee benefit plans registered on Form S-8 or any similar successor form, (iii) of equity securities registered to effect the acquisition of, or combination with, another Person on FormS-4 or any similar successor form, (iv) relating solely to the sale of non-convertible debt instruments and (v) of securities registered in connection with any dividend reinvestment plan.
 - "FINRA" has the meaning given such term in Section 2.5(xvi) herein.
- "Holder" means (i) the Initial Holder and (ii) any direct or indirect transferee of the Initial Holder who shall become a party to this Agreement in accordance with Section 2.9 and has agreed in writing to be bound by the terms of this Agreement.
 - "Initial Holder" has the meaning given such term in the introductory paragraph of this Agreement.
 - "Inspectors" has the meaning given such term in Section 2.5(xii) herein.
 - "Losses" has the meaning given such term in Section 2.8.1 herein.
 - "Marketed Underwritten Offering" has the meaning given such term in Section 2.1.3 herein.
 - "Permitted Transferee" has the meaning given such term in Section 2.9 herein.

"Person" or "person" means any individual, corporation, partnership, limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or government or other agency or political subdivision thereof.

"Piggyback Registration" has the meaning given such term in Section 2.3.1 herein.

"Records" has the meaning given such term in Section 2.5(xii) herein.

"register," "registered" and "registration" refer to a registration effected by preparing and filing a registration statement in compliance with the Securities Act, and the declaration or ordering of the effectiveness of such registration statement under the Securities Act (to the extent such declaration or order is required in order for such registration statement to become effective).

"Registrable Securities" means any shares of Company Common Stock issuable or issued upon the exercise of any Warrant and any shares of Company Common Stock issuable or issued upon the exercise of any Additional Warrant Securities; provided, however, that Registrable Securities shall not include shares of Company Common Stock (a) when a registration statement with respect to the sale of such shares of Company Common Stock has become effective under the Securities Act and such shares of Company Common Stock have been disposed of in accordance with such registration statement; (b) that have been sold to the public pursuant to Rule 144 or other exemption from registration under the Securities Act; (c) that have been otherwise transferred, new certificates for them not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of them shall not require registration or qualification of them under the Securities Act or any similar state law then in force; (d) as to which the Company has delivered an opinion of counsel reasonably satisfactory to the transfer agent for the Company Common Stock to the effect that such Registrable Securities are able to be sold by the Holders without restriction as to volume or manner of sale pursuant to Rule 144; (e) that are otherwise sold or transferred by a Holder in a transaction where its rights under this Agreement are not assigned; or (f) that have ceased to be outstanding.

"Requesting Holders" shall mean any Holder(s) requesting to have its (their) Registrable Securities included in any Demand Registration or Shelf Registration.

"Required Filing Date" has the meaning given such term in Section 2.2.1(b) herein.

"Rule 144" means Rule 144 under the Securities Act or any successor rule thereto.

"SEC" means the United States Securities and Exchange Commission or any other federal agency at the time administering the Securities Act.

"Securities Act" means the Securities Act of 1933, as amended, or any similar federal statute, and the rules and regulations promulgated by the SEC thereunder.

"Seller Affiliates" has the meaning given such term in Section 2.8.1 herein.

"Shelf Registration Statement" has the meaning given such term in Section 2.1.1 herein.

"Shelf Takedown" has the meaning given such term in Section 2.2.2(b) herein.

"Suspension Notice" has the meaning given such term in Section 2.6 herein.

"<u>Underwritten Offering</u>" shall mean an offering registered under the Securities Act in which securities of the Company are sold to one or more underwriters on a firm-commitment basis for reoffering to the public.

"Warrant" shall mean the warrant to purchase common stock, issued by the Company to the Initial Holder on or about the date hereof, together with the common stock purchase warrant in favor of the Initial Holder dated May 29, 2020.

1.2 Rules of Construction. Unless the context otherwise requires:

- (1) a term has the meaning assigned to it;
- (2) "or" is not exclusive;
- (3) words in the singular include the plural, and words in the plural include the singular;
- (4) whenever the masculine is used in this Agreement, the same shall include the feminine and whenever the feminine is used herein, the same shall include the masculine, where appropriate;
- (5) provisions apply to successive events and transactions; and
- (6) "herein," "hereof" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision.

ARTICLE 2 REGISTRATION RIGHTS

2.1 Shelf Registration.

- 2.1.1 Registration Requirement. The Company shall prepare and file a resale registration statement on FormS-3 under the Securities Act (it being agreed that such registration statement shall be a registration statement filed for an offering to be made on a delayed or continuous basis pursuant to Rule 415 (or any successor rule), including any post-effective amendment thereto, if then available to the Company, and if such Form S-3 is not then available to the Company, such resale registration statement shall be on Form S-1 or any similar or successor to such form under the Securities Act (the registration statement filed pursuant to this Section 2.1.1 being referred to as a "Shelf Registration Statement")) for the resale of all of the Registrable Securities as promptly as practicable after the date hereof, but in no event more than 90 days after the date hereof.
- 2.1.2 Effectiveness of the Registration Statement. The Company shall use its best efforts to cause the Shelf Registration Statement to be declared effective by the SEC staff no later than 180 days after the date hereof. Thereafter, the Company shall use its best efforts to keep such Shelf Registration Statement continuously effective, including by filing any necessary post-effective amendments to such Shelf Registration Statement or a new Shelf Registration Statement,

until the earlier of (x) the date on which all Registrable Securities have been sold pursuant to such Shelf Registration Statement or another Shelf Registration Statement filed under the Securities Act (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder) and (y) such time as the Registrable Securities are no longer outstanding or otherwise no longer constitute Registrable Securities. A Holder shall provide notice to the Company prior to any use of the Shelf Registration Statement by such Holder, and shall provide the information required by, and comply with the obligations under, Section 2.6 following the receipt of a Suspension Notice. Further, each Holder agrees to complete and execute all questionnaires and other documents reasonably required by the Company in order to prepare and file any Shelf Registration Statement.

2.1.3 Shelf Takedowns.

(a) Subject to the provisions of Section 2.1.3(b) hereof, any Holder or Holders of Registrable Securities shall be entitled, at any time and from time to time when a Shelf Registration Statement is effective, to sell such Registrable Securities held by such Holder or Holders as are then registered pursuant to a Shelf Registration Statement (each, a "Shelf Takedown"). The number of Shelf Takedowns that such Holder or Holders may effect pursuant to this Section 2.1.3 shall not be limited, provided, that the number of offerings where the plan of distribution contemplates a customary "road show" (including an "electronic road show") or other substantial marketing effort of by the Company and the underwriters (any such Underwritten Offering, a "Marketed Underwritten Offering") that may be effected hereunder shall be limited to a total of two (less any Demand Requests pursuant to Section 2.2.1), and such other restriction as may be set forth in Section 2.1.3(b) are complied with. Any such Shelf Takedown may be made in the United States by and pursuant to any method or combination of methods legally available to any Holder or Holders of Registrable Securities (including, but not limited to, an Underwritten Offering, a direct sale to purchasers, a sale to or through brokers, dealers or agents, a sale over the internet, Block Sales, derivative transactions with third parties, sales in connection with short sales and other hedging transactions). The Company shall comply with the applicable provisions of the Securities Act with respect to the disposition of all Registrable Securities covered by the Shelf Registration Statement in accordance with the intended methods of disposition by the Holder or Holders of Registrable Securities. If any Holder intends to sell any Registrable Securities pursuant to a Shelf Takedown, such Holder shall give the Company written notice of the consummation of each Shelf Takedown (whether or not such Shelf Takedown constitutes an Underwritten Offering) reasonably promptly after the consummatio

(b) Upon receipt of prior written notice by any Holder or Holders of Registrable Securities that it intends to effect a Shelf Takedown, the Company shall use its reasonable best efforts to cooperate in such Shelf Takedown, whether or not such Shelf Takedown constitutes an Underwritten Offering, by amending or supplementing the prospectus related to such Shelf Registration Statement as may be reasonably requested by such Holder or Holders for so long as such Holder or Holders holds Registrable Securities; provided, that the Company shall not be obligated to cooperate in an Underwritten Offering to be effected by means of a Block Sale if notice of such Underwritten Offering has not been delivered to the Company at least five Business Days prior to the intended launch of such Block Sale.

2.1.4 Selection of Underwriters. At the request of a majority of the Holders, the offering of Registrable Securities pursuant to a Shelf Takedown, shall be in the form of a "firm commitment" Underwritten Offering. In the case of an Underwritten Offering, a majority of such Holders shall select the investment banking firm or firms to manage the Underwritten Offering; provided, that such selection shall be subject to the prior consent of the Company, which consent shall not be unreasonably withheld, conditioned or delayed. No Holder may participate in any such Underwritten Offering unless such Holder (x) agrees to sell such Holder's Registrable Securities on the basis provided in any underwriting arrangements described above and (y) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements; provided, however, that any such Holder's representations and warranties in connection with any such registration shall be substantially consistent in substance and scope with those that are customarily made by selling securityholders to underwriting arrangements shall be several, not joint and several, among such Holders selling Registrable Securities, and the liability of each such Holder will be in proportion thereto; provided, further, that such liability will be limited to the net amount received by such Holder from the sale of such Holder's Registrable Securities pursuant to such Underwritten Offering.

2.1.5 [Reserved]

2.1.6 Deferral of Filing. If the filing, initial effectiveness or continued use of a Registration Statement, including a Shelf Registration Statement, filed hereunder would require the Company to make a public disclosure of material non-public information, which disclosure in the good-faith judgment of the Company based on the advice of counsel (i) would be required to be made in any registration statement so that such registration statement would not be materially misleading, (ii) would not be required to be made at such time but for the filing, effectiveness or continued use of such registration statement or (iii) would reasonably be expected to adversely affect in any material respect the Company or its business or the Company's ability to effect a *bona fide* material proposed acquisition, disposition, financing, reorganization, recapitalization or similar transaction, then the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement; provided that the Company shall not be permitted to do so (x) more than once in any six-month period or (y) for any single period of time in excess of 90 days, or for periods exceeding, in the aggregate, 90 days during any 12-month period. In the event that the Company exercises its rights under the preceding sentence, the Holders agree to suspend, promptly upon receipt of the notice referred to above, the use of any prospectus relating to such registration in connection with any sale or offer to sell Registrable Securities. In order to defer the filing of a registration statement pursuant to this Section 2.1.6, the Company shall promptly (but in any event within 10 days), upon determining to seek such deferral, deliver to each Requesting Holder a certificate signed by an executive officer of the Company stating that the Company is deferring such filing pursuant to this Section 2.1.6 and a statement of the reason for such deferral and an approximation of the antici

2.1.7 Form S-3. The Company shall use its reasonable best efforts to cause the Shelf Registration Statement to be registered on FormS-3 (or any successor form), and if the Company is not then eligible under the Securities Act to use Form S-3, such Shelf Registration Statement shall be registered on the form for which the Company then qualifies.

2.2 Demand Registration.

2.2.1 Request for Registration.

- (a) If the Company is unable to file, cause to become effective or maintain the effectiveness of a Shelf Registration Statement as required under Section 2.1, or at any time after the issuance of any Additional Warrant Securities, the Holder shall have the right to require the Company to, pursuant to the terms of this Agreement, register under and in accordance with the provisions of the Securities Act all or part of its or their Registrable Securities (a "Demand Registration"), by delivering to the Company written notice stating that such right is being exercised, naming, if applicable, the Holders whose Registrable Securities are to be included in such registration (collectively, the "Demanding Shareholders"), specifying the number of each such Demanding Shareholder's Registrable Securities to be included in such registration and, subject to Section 2.2.3 hereof, describing the intended method of distribution thereof (a "Demand Request").
- (b) Subject to this <u>Section 2.2.1</u> and <u>Section 2.2.5</u>, the Company shall file a registration statement in respect of a Demand Registration as soon as reasonably practicable and, in any event, within 30 days after receiving a Demand Request (the "<u>Required Filing Date</u>") and shall use reasonable best efforts to cause the same to be declared effective by the SEC as promptly as reasonably practicable after such filing; <u>provided</u>, <u>however</u>, that the Company shall not be obligated to effect:
 - (i) a Demand Registration pursuant to <u>Section 2.2.1(a)</u> within 90 days after the effective date of a previous Demand Registration or any previous registration statement in which the Holder or Holders of Registrable Securities was given piggyback rights pursuant to <u>Section 2.3</u> in which there was no reduction in the number of Registrable Securities to be included, and in each case in which the sale of Registered Securities was consummated; and
 - (ii) any Demand Registration if a Shelf Registration Statement is then effective, and such Shelf Registration Statement may be utilized by the Holder or Holders of Registrable Securities for the resale of Registrable Securities, including through an Underwritten Offering, without a requirement under the SEC's rules and regulations for a post-effective amendment thereto.

Notwithstanding the foregoing, the Company shall not be obligated to effect, in total, more than four Demand Registrations (less the number of any Shelf Takedowns constituting an Underwritten Offering), which may consist of (a) no more than two Demand Registrations where the plan of distribution contemplates a Marketed Underwritten Offering, less the number of any Shelf Takedowns constituting a Marketed Underwritten Offering and (b) no more than two Demand Registrations (less the number of any Shelf Takedowns constituting an Underwritten Offering) during any 12-month period.

(c) Each Holder requesting a Demand Registration agrees to complete and execute all questionnaires and other documents reasonably required by the Company in order to prepare and file any Shelf Registration Statement.

- 2.2.2 Selection of Underwriters. At the request of a majority of the Requesting Holders, the offering of Registrable Securities pursuant to a Demand Registration, shall be in the form of a "firm commitment" Underwritten Offering. In the case of an Underwritten Offering, a majority of the Requesting Holders shall select the investment banking firm or firms to manage the Underwritten Offering; provided, that such selection shall be subject to the prior consent of the Company, which consent shall not be unreasonably withheld, conditioned or delayed. No Holder may participate in any such Underwritten Offering unless such Holder (x) agrees to sell such Holder's Registrable Securities on the basis provided in any underwriting arrangements described above and (y) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements; provided, however, that any such Holder's representations and warranties in connection with any such registration shall be substantially consistent in substance and scope with those that are customarily made by selling securityholders to underwriters and issuers in underwritten offerings; provided, further, however, that the obligation of such Holder to indemnify pursuant to any such underwriting arrangements shall be several, not joint and several, among such Holders selling Registrable Securities, and the liability of each such Holder will be in proportion thereto; provided, further, that such liability will be limited to the net amount received by such Holder from the sale of such Holder's Registrable Securities pursuant to such Underwritten Offering.
- 2.2.3 Rights of Nonrequesting Holders. Upon receipt of any Demand Request, the Company shall promptly (but in any event within 10 days) give written notice of such proposed Demand Registration to all other Holders (if any), who shall have the right, exercisable by written notice to the Company within 15 days of their receipt of the Company's notice, to elect to include in such Demand Registration such portion of their Registrable Securities as they may request. All Holders requesting to have their Registrable Securities included in a Demand Registration in accordance with the preceding sentence shall be deemed to be "Requesting Holders" for purposes of this Section 2.2.
- 2.2.4 <u>Priority on Demand Registrations</u>. No securities to be sold for the account of any Person (including the Company) other than any Requesting Holder shall be included in a Demand Registration unless the managing underwriter or underwriters shall advise such Requesting Holder (or, in the case of a Demand Registration that is not an Underwritten Offering, such Requesting Holder determines in good faith after considering the relevant facts and circumstances at the relevant time) that the inclusion of such securities will not cause an Adverse Effect. Furthermore, if the managing underwriter or underwriters shall advise the Requesting Holder (or such Requesting Holder determines, as applicable, in good faith after considering the relevant facts and circumstances at the relevant time) that, even after exclusion of all securities of other Persons pursuant to the immediately preceding sentence, the amount of securities proposed to be included in such Demand Registration by the Requesting Holders is sufficiently large to cause an Adverse Effect, the number of securities to be included in such Demand Registration shall equal the number of shares which the Requesting Holder is so advised can be sold in such offering without an Adverse Effect, allocated *pro rata* among the Requesting Holders on the basis of the number of Registrable Securities requested to be included in such registration by each such Requesting Holder.

- 2.2.5 <u>Deferral of Filing</u>. With respect to any Demand Registration, the obligation of the Company to file, accelerate the initial effectiveness or continue the effectiveness of a registration statement shall be limited to the extent set forth in <u>Section 2.1.6</u>. If the Company so postpones the filing of a prospectus or the effectiveness of a registration statement with respect to a Demand Registration for the reasons set forth in <u>Section 2.1.6</u>, the Holders shall be entitled to withdraw such request and, if such request is withdrawn, such registration request shall not count for the purposes of the limitations set forth in <u>Section 2.2.2</u>. The Company shall promptly give the Holders requesting registration thereof pursuant to this <u>Section 2</u> written notice of any postponement made in accordance with the preceding sentence. A deferral of the filing of a registration statement pursuant to this <u>Section 2.2.5</u>, shall be lifted, and the requested registration statement shall be filed forthwith. In order to defer the filing of a registration statement pursuant to this <u>Section 2.2.5</u>, the Company shall promptly (but in any event within 10 days), upon determining to seek such deferral, deliver to each Requesting Holder a certificate signed by an executive officer of the Company stating that the Company is deferring such filing pursuant to this <u>Section 2.2.5</u> and a statement of the reason for such deferral and an approximation of the anticipated delay. Within 20 days after receiving such certificate, the holders of a majority of the Registrable Securities held by the Requesting Holders and for which registration was previously requested may withdraw such Demand Request by giving notice to the Company; if withdrawn, the Demand Request shall be deemed not to have been made for all purposes of this Agreement.
- 2.2.6 Form S-3. The Company shall use its reasonable best efforts to cause Demand Registrations to be registered on FormS-3 (or any successor form), and if the Company is not then eligible under the Securities Act to use Form S-3, Demand Registrations shall be registered on the form for which the Company then qualifies.

2.3 Piggyback Registrations.

2.3.1 Right to Piggyback. Each time the Company proposes to register any of its equity securities (other than pursuant to an Excluded Registration) under the Securities Act for sale to the public (whether for the account of the Company or the account of any securityholder of the Company) (a "Piggyback Registration"), the Company shall give prompt written notice to each Holder of Registrable Securities (which notice shall be given not less than 10 days prior to the anticipated filing date of the Company's registration statement), which notice shall offer each such Holder the opportunity to include any or all of such Holder's Registrable Securities in such registration statement on the same terms and conditions as the same class of securities otherwise being sold pursuant to such registration statement, subject to the limitations contained in Section 2.3.2 hereof. Each Holder who desires to have such Holder's Registrable Securities included in such registration statement shall so advise the Company in writing (stating the number of shares desired to be registered) within five days after the date of such notice from the Company. Any Holder shall have the right to withdraw such Holder's request for inclusion of such Holder's Registrable Securities in any registration statement pursuant to this Section 2.3.1 by giving written notice to the Company of such withdrawal on or before the fifth day prior to the planned effective date of such Piggyback Registration. Subject to Section 2.3.2 below, the Company shall include in such registration statement all such Registrable Securities so requested to be included therein;

provided, however, that the Company may at any time, in its sole discretion and without the consent of the Holders, delay, withdraw or cease proceeding with any such registration if it shall at the same time withdraw or cease proceeding with the registration of all other equity securities originally proposed to be registered and will have no liability to the Holder in connection with such termination or withdrawal, except for the obligation to pay any registration expenses pursuant to Section 2.7.3.

2.3.2 Priority on Piggyback Registrations.

(a) If a Piggyback Registration is an Underwritten Offering and was initiated by the Company, and if the managing underwriter (or in the case of a Piggyback Registration that is not an Underwritten Offering, the Company, in good faith) advises the Company that the inclusion of Registrable Securities requested to be included in the Registration Statement would cause an Adverse Effect, the Company shall include in such registration statement (i) first, the securities the Company proposes to sell, (ii) second, the Registrable Securities of any Holder requested to be included in such registration, *pro rata* among the Holders of such Registrable Securities on the basis of the number of Registrable Securities owned by each such Holder and (iii) third, any other securities requested to be included in such registration; provided, that if such other securities have been requested to be included pursuant to a registration rights agreement, then such securities would be included as set forth in (ii) above as if they were Registrable Securities of a Holder. If, as a result of the provisions of this Section 2.3.2(a), any Holder shall not be entitled to include all Registrable Securities in a registration that such Holder has requested to be so included, such Holder may withdraw such Holder's request to include Registrable Securities in such registration statement on or before the fifth day prior to the planned effective date of such Piggyback Registration.

(b) If a Piggyback Registration is an Underwritten Offering and was initiated by a securityholder of the Company, and if the managing underwriter (or in the case of a Piggyback Registration that is not an Underwritten Offering, the Company, in good faith) advises the Company that the inclusion of Registrable Securities requested to be included in the Registration Statement would cause an Adverse Effect, the Company shall include in such registration statement (i) first, the securities of the Existing Registration Rights Holders requested to be included in such registration, (ii) second, the securities requested to be included therein by the securityholders requesting such registration, (iii) third, the Registrable Securities requested to be included in such registration by any Holder, *pro rata* among the Holders on the basis of the number of Registrable Securities owned by each such Holder and (iv) fourth, any other securities requested to be included in such registration (including securities to be sold for the account of the Company). If, as a result of the provisions of this Section 2.3.2(b), any Holder shall not be entitled to include all Registrable Securities in a registration that such Holder has requested to be so included, such Holder may withdraw such Holder's request to include Registrable Securities in such registration statement on or before the fifth day prior to the planned effective date of such Piggyback Registration.

(c) No Holder may participate in any registration statement in respect of a Piggyback Registration hereunder unless such Holder (x) agrees to sell such Holder's Registrable Securities on the basis provided in any underwriting arrangements approved by the Company, in the case of an Underwritten Offering and (y) completes and executes all

questionnaires, powers of attorney, indemnities, underwriting agreements and other documents, each in customary form, reasonably required under the terms of such underwriting arrangements; provided, however, that any such Holder's representations and warranties in connection with any such registration shall be of a substance and scope as are customarily made by selling securityholders to underwriters and issuers in underwritten offerings; provided, further, however, that the obligation of such Holder to indemnify pursuant to any such underwriting arrangements shall be several, not joint and several, among such Holders selling Registrable Securities, and the liability of each such Holder will be in proportion thereto; provided, further, that such liability will be limited to the net amount received by such Holder from the sale of such Holder's Registrable Securities pursuant to such registration.

2.4 Holdback Agreements.

- (a) In the case of any Underwritten Offering by any Holder hereunder, the Company shall not effect any public sale or distribution of its equity securities, or any securities convertible into or exchangeable or exercisable for such securities, during the seven days prior to and during the 30-day period beginning on the effective date of any registration statement filed in connection with such Underwritten Offering or, in the case of an Underwritten Offering pursuant to a Shelf Takedown, the filing of any prospectus relating to the offer and sale of Registrable Securities (or, in either case, such shorter period that any lock-up period with respect to such Underwritten Offering is in effect), except (i) pursuant to any registrations on FormS-4 or Form S-8 or any successor form, (ii) pursuant to any registrations filed in connection with an exchange offer or any employee benefit or dividend reinvestment plan or (iii) unless the underwriters managing any such Underwritten Offering otherwise agree. The underwriters in connection with such Underwritten Offering are intended third-party beneficiaries of this Section 2.4(a) and shall have the right and power to enforce the provisions hereof as though they were a party thereto.
- (b) Each Holder agrees, in the event of an Underwritten Offering by the Company (whether for the account of the Company or otherwise), not to offer, sell, contract to sell or otherwise dispose of any Registrable Securities, or any securities convertible into or exchangeable or exercisable for such securities, including any sale pursuant to Rule 144 (except as part of such Underwritten Offering), during the seven days prior to, and during the 30-day period beginning on, the effective date of the registration statement for such Underwritten Offering (or, in the case of an offering pursuant to an effective shelf registration statement pursuant to Rule 415, the pricing date for such Underwritten Offering) (or, in either case, such shorter period that any lock-up period with respect to such Underwritten Offering is in effect). The underwriters in connection with such Underwritten Offering are intended third-party beneficiaries of this Section 2.4(b) and shall have the right and power to enforce the provisions hereof as though they were a party thereto.
- 2.5 Registration Procedures. If and whenever any Holder has requested that any Registrable Securities be registered pursuant to this Agreement, the Company will use its reasonable best efforts to effect the registration and the sale of such Registrable Securities in accordance with the intended method of disposition thereof as promptly as is reasonably practicable, and pursuant thereto the Company will as expeditiously as possible:

- (i) prepare and file with the SEC, pursuant to Section 2.2.1(b) with respect to any Demand Registration, a registration statement on any appropriate form under the Securities Act with respect to such Registrable Securities and use its reasonable best efforts to cause such registration statement to become effective; provided, that as far in advance as practicable before filing such registration statement or any amendment thereto, the Company will furnish to the selling Holders copies of reasonably complete drafts of all such documents prepared to be filed (including exhibits), and any such Holder shall have the opportunity to review and reasonably object, as promptly as is reasonably practicable, to any information contained therein and the Company will make corrections reasonably requested by such Holder with respect to such information prior to filing any such registration statement or amendment; provided, that the Company shall not have any obligation to modify any information if the Company reasonably believes in good faith that so doing would cause (i) the registration statement to contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading or (ii) the prospectus to contain an untrue statement of a material fact or to omit to state a material fact necessary in order to make the statements made, in light of the circumstances under which they were made, not misleading;
- (ii) except in the case of a Shelf Registration Statement, prepare and file with the SEC such amendments, post-effective amendments, and supplements to such registration statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective for a period of not less than 180 days (or such lesser period as is necessary for the underwriters in an underwritten offering to sell unsold allotments) and comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement during such period in accordance with the intended methods of disposition by the sellers thereof set forth in such registration statement;
- (iii) in the case of a Shelf Registration Statement, prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities subject thereto for a period ending on the earlier of (x) 24 months after the effective date of such registration statement, (y) the date when all restrictive legends on the Registrable Securities have been removed or (z) the date on which all the Registrable Securities held by any Holder cease to be Registrable Securities;
- (iv) furnish to each seller of Registrable Securities and the underwriters of any Underwritten Offering, such number of copies of such registration statement, each amendment and supplement thereto, the prospectus included in such registration statement (including each preliminary prospectus), any prospectus supplement, any documents incorporated by reference therein and

such other documents as such seller or underwriters may reasonably request for purposes of permitting such seller's or underwriters' review in order to facilitate the disposition of the Registrable Securities owned by such seller or the sale of such securities by such underwriters (it being understood that, subject to Section 2.6 and the requirements of the Securities Act and applicable state securities laws, the Company consents to the use of the prospectus and any amendment or supplement thereto by each seller and the underwriters in connection with any Underwritten Offering covered by the registration statement of which such prospectus, amendment or supplement is a part);

(v) use its reasonable best efforts to register or qualify such Registrable Securities under such other securities or blue sky laws of such jurisdictions in the United States as the managing underwriter reasonably requests (or, in the event the registration statement does not relate to an Underwritten Offering, as the holders of a majority of such Registrable Securities may reasonably request); use its reasonable best efforts to keep each such registration or qualification (or exemption therefrom) effective during the period in which such registration statement is required to be kept effective and to take any other action that may be reasonably necessary or advisable to enable each seller to consummate the disposition of the Registrable Securities owned by such seller in such jurisdictions (provided, however, that the Company will not be required to (A) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subparagraph, (B) subject itself to taxation in any jurisdiction wherein it is not so subject or (C) take any action that would subject it to general service of process in any jurisdiction where it is not then so subject);

(vi) promptly notify each seller and each underwriter of any Underwritten Offering and (if requested by any such Person) confirm such notice in writing (A) when a prospectus or any prospectus supplement or post-effective amendment has been filed and, with respect to a registration statement or any post-effective amendment, when the same has become effective, (B) of the issuance by any state securities or other regulatory authority of any order suspending the qualification or exemption from qualification of any of the Registrable Securities under state securities or "blue sky" laws or the initiation of any proceedings for that purpose, and (C) of the happening of any event which makes any material statement made in a registration statement or related prospectus untrue or which requires the making of any material changes in such registration statement, prospectus or documents so that they will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and, as promptly as practicable thereafter, prepare and file with the SEC and furnish a supplement or amendment to such prospectus so that, as thereafter deliverable to the purchasers of such Registrable Securities, such prospectus will not contain any untrue statement of a material fact or omit a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading;

- (vii) permit any selling Holder, which in such Holder's judgment, based on the advice of counsel, might reasonably be deemed to be an underwriter or a controlling person of the Company, to participate in the preparation of such registration or comparable statement, to the extent necessary, and to require the insertion therein of material, furnished to the Company in writing, which in the reasonable judgment of such Holder and its counsel should be included; provided, that the Company shall not have any obligation to include such information if the Company reasonably believes in good faith that so doing would cause (i) the registration statement to contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading or (ii) the prospectus to contain an untrue statement of a material fact or to omit to state a material fact necessary in order to make the statements made, in light of the circumstances under which they were made, not misleading;
- (viii) in the case of any Underwritten Offering, make reasonably available members of management of the Company, as selected by the Holders of a majority of the Registrable Securities included in such registration, for assistance in the selling effort relating to the Registrable Securities covered by such registration, including, but not limited to, the participation of such members of the Company's management in road show presentations as the underwriters reasonably request; <u>provided</u>, that the underwriter shall take into account the reasonable business requirements of the Company in determining the scheduling and duration of any road show;
- (ix) otherwise use its reasonable best efforts to comply with the Securities Act, the Exchange Act and all other applicable rules and regulations of the SEC, and make generally available to the Company's securityholders an earnings statement satisfying the provisions of Section 11(a) of the Securities Act no later than 45 days after the end of the 12-month period beginning with the first day of the Company's first fiscal quarter commencing after the effective date of a registration statement, which earnings statement shall cover said 12-month period, and which requirement will be deemed to be satisfied if the Company timely files complete and accurate information on Forms 10-Q, 10-K and 8-K under the Exchange Act and otherwise complies with Rule 158 under the Securities Act;
- (x) if requested by the managing underwriter of any Underwritten Offering or any seller, promptly incorporate in a prospectus supplement or post-effective amendment such information as the managing underwriter or such seller reasonably requests to be included therein, including, without limitation, with respect to the Registrable Securities being sold by such seller, the purchase price being paid therefor by the underwriters and with respect to any other terms of the underwritten offering of the Registrable Securities to be sold in such offering, and promptly make all required filings of such prospectus supplement or post-effective amendment;

(xi) cooperate with the seller and the managing underwriter of any Underwritten Offering to facilitate the timely preparation and delivery of certificates (which shall not bear any restrictive legends unless required under applicable law) representing securities sold under any registration statement, and enable such securities to be in such denominations and registered in such names as the managing underwriter or such sellers may request as promptly as reasonably practicable prior to any sale of Registrable Securities and keep available and make available to the Company's transfer agent prior to the effectiveness of such registration statement a supply of such certificates;

(xii) in the case of an Underwritten Offering, upon reasonable notice and during normal business hours, make reasonably available for inspection by any seller, any underwriter participating in any disposition pursuant to any registration statement, and any attorney, accountant or other agent or representative retained by any such seller or underwriter (collectively, the "Inspectors"), relevant financial and other records, pertinent corporate documents and properties of the Company (collectively, the "Records"), as shall be reasonably necessary to enable them to exercise their due diligence responsibility, and cause the Company's officers, directors and employees to supply information reasonably requested by any such Inspector in connection with such registration statement; provided, however, that, unless the disclosure of such Records is necessary to avoid or correct a misstatement or omission in the registration statement or the release of such Records is ordered pursuant to a subpoena or other order from a court of competent jurisdiction, the Company shall not be required to provide any information under this subparagraph (xii) if (A) the Company believes, after consultation with counsel for the Company, that either (1) the requested Records constitute confidential commercial and/or supervisory information within the meaning of 5 U.S.C. § 552(b)(4) and (8), respectively, or (2) to do so would cause the Company to forfeit an attorney-client privilege that was applicable to such information, or (B) if the Company has requested and been granted from the SEC confidential treatment of such information contained in any filing with the SEC or documents provided supplementally or otherwise; provided, further, however, that any Records and other information provided under this Section 2.5(xii) that is not generally publicly available shall be subject to such confidential treatment as is customary for underwriters' due diligence reviews;

(xiii) in the case of any Underwritten Offering, use its reasonable best efforts to furnish to each seller and the underwriter a signed counterpart of (A) an opinion or opinions of counsel to the Company (and/or internal counsel if acceptable to the managing underwriters and the sellers), and (B) a comfort letter or comfort letters from the Company's independent public accountants, each in customary form and covering such matters of the type customarily covered by opinions or comfort letters, as the case may be, as the seller or managing underwriter reasonably requests;

- (xiv) use its reasonable best efforts to cause the Registrable Securities covered by any registration statement to be listed on the primary national securities exchange, if any, on which similar securities issued by the Company are then listed;
 - (xv) provide a transfer agent and registrar for all Registrable Securities registered hereunder;
- (xvi) reasonably cooperate with each seller and each underwriter of any Underwritten Offering participating in the disposition of such Registrable Securities and their respective counsel in connection with any filings required to be made with the Financial Industry Regulatory Authority, Inc. ("FINRA");
- (xvii) during the period when the prospectus is required to be delivered under the Securities Act, promptly file all documents required to be filed with the SEC pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act;
- (xviii) notify each seller of Registrable Securities promptly of any request by the SEC for the amending or supplementing of any registration statement or prospectus relating to such seller's Registrable Securities;
- (xix) enter into such agreements (including underwriting agreements) as are customary in connection with an Underwritten Offering; and
- (xx) advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the SEC suspending the effectiveness of a registration statement relating to the such seller's Registrable Securities or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal as soon as practicable if such stop order should be issued.

The Company may, from time to time, require any Holder of Registrable Securities as to which any registration is being effected to furnish to the Company in writing such information as the Company reasonably determines, based on the advice of counsel, is required or advised to be included in connection with such registration regarding such Holder and the distribution of such Registrable Securities, and the Company may exclude from such registration the Registrable Securities of such Holder if such Holder fails to furnish such information within 15 days of receiving such request.

2.6 <u>Suspension of Dispositions</u>. Each Holder agrees by acquisition of any Registrable Securities that, upon receipt of any notice (a <u>Suspension Notice</u>") from the Company of the happening of any event of the kind described in <u>Section 2.5(vi)(C)</u>, such Holder will forthwith discontinue disposition of Registrable Securities until such Holder's receipt of the copies of the supplemented or amended prospectus, or until it is advised in writing (the "<u>Advice</u>") by the Company that the use of the prospectus may be resumed, and has received copies of any additional or supplemental filings which are incorporated by reference in the prospectus, and, if so directed by the Company, such Holder will deliver to the Company all copies, other than permanent file copies then in such Holder's possession, of the prospectus covering such Registrable Securities

current at the time of receipt of such notice. In the event the Company shall give any such notice, the time period regarding the effectiveness of registration statements set forth in Section 2.5(ii) and Section 2.5(iii) hereof shall be extended by the number of days during the period from and including the date of the giving of the Suspension Notice to and including the date when each seller of Registrable Securities covered by such registration statement shall have received the copies of the supplemented or amended prospectus or the Advice. The Company shall use its reasonable best efforts and take such actions as are reasonably necessary to render the Advice as promptly as reasonably practicable. In any event, the Company shall not be entitled to deliver more than two Suspension Notices in any one year.

2.7 Registration Expenses.

2.7.1 Shelf Registration. The Company shall be responsible for all reasonable and documented, out-of-pocket fees and expenses incident to any Shelf Registration Statement or Shelf Takedown including, without limitation, the Company's performance of or compliance with this Article 2, all registration and filing fees, all fees and expenses associated with filings required to be made with FINRA (including, if applicable, the reasonable fees and expenses of any "qualified independent underwriter" as such term is defined in FINRA Rule 5121, and of its counsel), as may be required by the rules and regulations of FINRA, fees and expenses of compliance with securities or "blue sky" laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the Registrable Securities), rating agency fees, printing expenses (including expenses of printing certificates for the Registrable Securities and of printing prospectuses if the printing of prospectuses is requested by any Holder of Registrable Securities), messenger and delivery expenses, the fees and expenses incurred in connection with any listing or quotation of the Registrable Securities, and the fees and expenses of counsel for the Company and its independent certified public accountants (including the expenses of any special audit or "comfort letters" required by or incident to such performance); provided, however, that any underwriting discounts, commissions, or fees attributable to the sale of the Registrable Securities will be borne by the Holders pro rata on the basis of the number of shares so sold and the fees and expenses of any counsel, accountants, or other persons retained or employed by any Holder and any applicable transfer taxes will be borne by such Holder. The Company shall be responsible for the reasonable and documented fees and expenses of one firm of attorneys retained by all of the Holders in the aggregate in connection with a Shelf Registration Statement and the sale of Registrable Securities in a Shelf Takedown. Notwithstanding the foregoing, the Company shall not be responsible for the fees and expenses of any additional counsel, or any of the accountants, agents or experts retained by the Holders in connection with the sale of Registrable Securities in a Shelf Takedown. The Company will also be responsible for its internal expenses in any Shelf Registration Statement and Shelf Takedown (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties, the expense of any annual audit and the expense of any liability insurance).

2.7.2 <u>Demand Registrations</u>. The Company shall be responsible for all reasonable and documented, out-of-pocket fees and expenses incident to any Demand Registration including, without limitation, the Company's performance of or compliance with this <u>Article 2</u>, all registration and filing fees, all fees and expenses associated with filings required to be made with FINRA (including, if applicable, the reasonable fees and expenses of any "qualified independent underwriter" as such term is defined in FINRA Rule 5121, and of its counsel), as may be required

by the rules and regulations of FINRA, fees and expenses of compliance with securities or "blue sky" laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the Registrable Securities), rating agency fees, printing expenses (including expenses of printing certificates for the Registrable Securities and of printing prospectuses if the printing of prospectuses is requested by any Holder of Registrable Securities), messenger and delivery expenses, the fees and expenses incurred in connection with any listing or quotation of the Registrable Securities, and the fees and expenses of counsel for the Company and its independent certified public accountants (including the expenses of any special audit or "comfort letters" required by or incident to such performance). The Holders shall be responsible for (i) any underwriting discounts, commissions, or fees attributable to the sale of the Registrable Securities, on a *pro rata* basis on the basis of the number of shares so sold whether or not any registration statement becomes effective, and (ii) any applicable transfer taxes. The Company shall be responsible for the reasonable and documented fees and expenses of one firm of attorneys retained by all of the Holders in the aggregate in connection with the sale of Registrable Securities in a Demand Registration. Notwithstanding the foregoing, the Company shall not be responsible for the fees and expenses of any additional counsel, or any of the accountants, agents or experts retained by the Holders in connection with the sale of Registrable Securities in a Demand Registration. The Company will also be responsible for its internal expenses in any Demand Registration (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties, the expense of any annual audit and the expense of any liability insurance).

2.7.3 Piggyback Registrations. All fees and expenses incident to any Piggyback Registration including, without limitation, the Company's performance of or compliance with this Article 2, all registration and filing fees, all fees and expenses associated with filings required to be made with FINRA (including, if applicable, the reasonable and documented fees and expenses of any "qualified independent underwriter" as such term is defined in FINRA Rule 5121, and of its counsel), as may be required by the rules and regulations of FINRA, fees and expenses of compliance with securities or "blue sky" laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the Registrable Securities), rating agency fees, printing expenses (including expenses of printing certificates for the Registrable Securities and of printing prospectuses), messenger and delivery expenses, the fees and expenses incurred in connection with any listing or quotation of the Registrable Securities, fees and expenses of counsel for the Company and its independent certified public accountants (including the expenses of any special audit or "comfort letters" required by or incident to such performance) and the fees and expenses of other persons retained by the Company, will be borne by the Company (unless paid by a securityholder that is not a Holder for whose account the registration is being effected) whether or not any registration statement becomes effective; provided, however, that any underwriting discounts, commissions, or fees attributable to the sale of the Registrable Securities will be borne by the Holders pro rata on the basis of the number of shares so sold and the fees and expenses of any counsel, accountants, or other persons retained or employed by any Holder and any applicable transfer taxes will be borne by such Holder.

2.8 Indemnification.

2.8.1 The Company agrees to indemnify and reimburse, to the fullest extent permitted by law, each seller of Registrable Securities, and each of its employees, advisors, agents, representatives, partners, officers, and directors and each Person who controls such seller (within the meaning of the Securities Act) and any agent or investment advisor thereof (collectively, the "Seller Affiliates") (A) against any and all losses, claims, damages, liabilities, and expenses, joint or several (including, without limitation, reasonable attorneys' fees and disbursements except as limited by Section 2.8.3) (collectively, "Losses") based upon, arising out of, related to or resulting from any untrue or alleged untrue statement of a material fact contained in any registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto, or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, (B) against any and all Losses, as incurred, to the extent of the aggregate amount reasonably paid in settlement of any litigation or investigation or proceeding by any governmental agency or body, commenced or threatened, or of any claim whatsoever based upon, arising out of, related to or resulting from any such untrue statement or omission or alleged untrue statement or omission made by the Company in any registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto; provided, that such settlement is effected with the consent of the Company (such consent not to be unreasonably withheld); and (C) against any Losses as may be reasonably incurred in investigating, preparing, or defending against any litigation, or investigation or proceeding by any governmental agency or body, commenced or threatened, or any claim whatsoever based upon, arising out of, related to or resulting from any such untrue statement or omission or alleged untrue statement or omission, or such violation of the Securities Act, the Exchange Act or any other similar federal or state securities laws or any rule or regulation promulgated thereunder, to the extent that any such expense or cost is not paid under subparagraph (A) or (B) above; provided, that the Company will have no obligation to provide any indemnification or reimbursement hereunder to the extent that any such Losses (or actions or proceedings in respect thereof) arise out of or are based upon an untrue statement or alleged untrue statement or omission or alleged omission made in such registration statement, prospectus, or any preliminary prospectus or any amendment thereof or supplement thereto, in reliance upon and in substantial conformity with information furnished in writing to the Company by such seller or any of its Seller Affiliates (or on such seller's or Seller Affiliate's behalf) for use therein.

2.8.2 In connection with any registration statement in which a seller of Registrable Securities is participating, each such seller will furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such registration statement or prospectus and, to the fullest extent permitted by law, each such seller will indemnify and reimburse, to the fullest extent permitted by law, the Company and each of its employees, advisors, agents, representatives, partners, officers and directors and each Person who controls the Company (within the meaning of the Securities Act or the Exchange Act) and any agent or investment advisor ("Company Indemnified Persons") thereof against any and all Losses resulting from any untrue statement or alleged untrue statement of a material fact contained in the registration statement, prospectus, or any preliminary prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or alleged untrue statement or omission or alleged omission is contained in any information or affidavit so furnished in writing to the Company by such seller or any Seller Affiliates (or on such seller's or Seller Affiliate's behalf) specifically for inclusion in the registration statement, prospectus, or any preliminary prospectus or any amendment thereof or supplement thereto and the Holders agree to reimburse the Company Indemnified Persons for any

legal or other expenses reasonably incurred by it in connection with investigating or defending any such action or claim as such expenses are incurred; provided, that the obligation to indemnify will be several, not joint and several, among such sellers of Registrable Securities, and the liability of each such seller of Registrable Securities will be in proportion and limited to the net amount received by such seller from the sale of Registrable Securities pursuant to such registration statement; provided, however, that such seller of Registrable Securities shall not be liable in any such case to the extent that prior to the filing of any such registration statement or prospectus or amendment thereof or supplement thereto, such seller has furnished in writing to the Company information expressly for use in such registration statement or prospectus or any amendment thereof or supplement thereto which corrected or made not misleading information previously furnished to the Company.

2.8.3 Any Person entitled to indemnification hereunder will (A) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided, that the failure to give such notice shall not limit the rights of such Person) and (B) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party; provided, however, that any person entitled to indemnification hereunder shall have the right to employ separate counsel and to participate in the defense of such claim, but the reasonable and documented fees and expenses of such counsel shall be at the expense of such person unless (X) the indemnifying party has agreed to pay such fees or expenses or (Y) the indemnifying party shall have failed to assume the defense of such claim and employ counsel reasonably satisfactory to such person. If such defense is not assumed by the indemnifying party as permitted hereunder, the indemnifying party will not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent will not be unreasonably withheld, conditioned or delayed). If such defense is assumed by the indemnifying party pursuant to the provisions hereof, such indemnifying party shall not settle or otherwise compromise the applicable claim unless (1) such settlement or compromise contains a full and unconditional release of the indemnified party or (2) the indemnified party otherwise consents in writing. An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim will not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party, a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim, in which event the indemnifying party shall be obligated to pay the reasonable fees and disbursements of such additional counsel or counsels.

2.8.4 Each party hereto agrees that, if for any reason the indemnification provisions contemplated by Section 2.8.1 or Section 2.8.2 are unavailable to or insufficient to hold harmless an indemnified party in respect of any Losses (or actions in respect thereof) referred to therein, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of such Losses (or actions in respect thereof) in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party in connection with the actions which resulted in the losses, claims, damages, liabilities or expenses as well as any other relevant equitable considerations. The relative fault of such indemnifying party and indemnified party shall be determined by reference to, among other things, whether the

untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact relates to information supplied by such indemnifying party or indemnified party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 2.8.4 were determined by pro rata allocation (even if the Holders or any underwriters or all of them were treated as one entity for such purpose) or by any other method of allocation which does not take account of the equitable considerations referred to in this Section 2.8.4. The amount paid or payable by an indemnified party as a result of Losses (or actions in respect thereof) referred to above shall be deemed to include any legal or other fees or expenses reasonably incurred by such indemnified party in connection with investigating or, except as provided in Section 2.8.3, defending any such action or claim. Notwithstanding the provisions of this Section 2.8.4, no Holder shall be required to contribute an amount greater than the dollar amount by which the net proceeds received by such Holder with respect to the sale of any Registrable Securities exceeds the amount of damages which such Holder has otherwise been required to pay by reason of any and all untrue or alleged untrue statements of material fact or omissions or alleged omissions of material fact made in any registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto related to such sale of Registrable Securities. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation. The Holders' obligations in this Section 2.8.4 to contribute shall be several in proportion to the amount of Registrable Securities registered by it and not

If indemnification is available under this <u>Section 2.8.</u>, the indemnifying parties shall indemnify each indemnified party to the full extent provided in <u>Section 2.8.1</u> and <u>Section 2.8.2</u> without regard to the relative fault of said indemnifying party or indemnified party or any other equitable consideration provided for in this <u>Section 2.8.4</u> subject, in the case of any Holder, to the limited dollar amounts set forth in<u>Section 2.8.2</u>.

- 2.8.5 The indemnification and contribution provided for under this Agreement will remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director, or controlling Person of such indemnified party and will survive the transfer of securities.
- 2.9 <u>Transfer of Registration Rights</u>. The rights of each Holder under this Agreement may be assigned or transferred to (i) any Affiliate of the Holder or (ii) third-party transferees of the Registrable Securities or Warrants that are not Affiliates of one another and that each acquire, or agree to acquire, an amount of Registrable Securities (or Warrants exercisable therefor), and, in the case of both (i) and (ii), such Affiliate of the Holder or transferee enters into a Joinder Agreement, substantially in the form of <u>Exhibit A</u> hereto (collectively, a "<u>Permitted Transferee</u>").

ARTICLE 3

- 3.1 Rule 144. The Company will file the reports required to be filed by it under the Exchange Act and the rules and regulations adopted by the SEC thereunder (or, if the Company is not required to file such reports, will, upon the request of the Holders, make publicly available other information so long as necessary to permit sales of Registrable Securities pursuant to Rule 144) and will take such further action as the Holders may reasonably request, all to the extent required from time to time to enable the Holders to sell Registrable Securities without registration under the Securities Act within the limitation of the exemption provided by Rule 144 or any similar rule or regulation hereafter adopted by the SEC. Upon the reasonable request of any Holder, the Company will deliver to such parties a written statement as to whether it has complied with such requirements. If any Initial Holder seeks to sell Company Common Stock under Rule 144, any legal opinion reasonably required by the transfer agent to effect such sale shall be provided by, or at the expense of, the Company.
- **3.2** <u>Preservation of Rights.</u> The Company will not (i) grant any registration rights to third parties which are more favorable than or inconsistent with the rights granted hereunder or (ii) enter into any agreement, take any action or permit any change to occur, with respect to its securities that violates or subordinates the rights expressly granted to the Holders; <u>provided, however</u>, that the Company may take any such actions as are necessary to memorialize or effect any existing arrangements or agreements for the grant of registrations rights to the Existing Registration Rights Holders.

ARTICLE 4 TERMINATION

4.1 <u>Termination</u>. This Agreement shall terminate and be of no further force and effect at the earliest to occur of (i) its termination by the written agreement of all parties or their respective successors in interest, (ii) with respect to any Holder, the date on which all Company Common Stock held by such Holder have ceased to be Registrable Securities, (iii) with respect to the Company, the date on which all Company Common Stock has ceased to be Registrable Securities and (iv) the dissolution, liquidation or winding up of the Company.

ARTICLE 5 MISCELLANEOUS

5.1 Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 5.1):

If to the Company:

Baudax Bio, Inc. 490 Lapp Road, Malvern, PA 19355 Attention Gerri Henwood, Chief Executive Officer Email: ghenwood@baudaxbio.com

with a copy to (which shall not constitute notice):

Goodwin Procter 2929 Arch Street, Suite 1700 Philadelphia, PA 19104 Attention: Jennifer Porter Email: <u>JPorter@goodwinlaw.com</u>

If to the Initial Holder:

One Bryant Park, 38th Floor New York, NY 10036 Attention: Evan Bedil Telephone: 212-500-3094 Email: ebedil@marathonfund.com

with a copy to (which shall not constitute notice):

Covington & Burling LLP
The New York Times Building
620 Eighth Avenue
New York, New York 10018
Attention: Peter Schwartz
Email: pschwartz@cov.com

Any notice or communication hereunder shall be deemed to have been given or made as of the date so delivered if personally delivered and five calendar days after mailing if sent by registered or certified mail (except that a notice of change of address shall not be deemed to have been given until actually received by the addressee). If a notice or communication is mailed in the manner provided above, it is duly given, whether or not the addressee receives it.

- **5.2** <u>Authority</u>. Each of the parties hereto represents to the other that (i) it has the corporate power and authority to execute, deliver and perform this Agreement, (ii) the execution, delivery and performance of this Agreement by it has been duly authorized by all necessary corporate action and no such further action is required, (iii) it has duly and validly executed and delivered this Agreement and (iv) this Agreement is a legal, valid and binding obligation, enforceable against it in accordance with its terms subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and general equity principles.
- 5.3 Governing Law. This Agreement shall be governed by and construed and interpreted in accordance with the laws of the State of New York irrespective of the choice of laws principles of the State of New York. Each party hereto irrevocably submits to the exclusive jurisdiction of the courts of the State of New York sitting in the County of New York or the United States District Court for the Southern District of New York and the appellate courts having jurisdiction of appeals in such courts to resolve any dispute, controversy or claim arising out of, or relating to, the transactions contemplated by this Agreement, or the validity, interpretation, breach or termination of any provision of this Agreement.

- 5.4 Waiver of Jury Trial. EACH PARTY HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY, EACH PARTY HERETO (I) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (II) ACKNOWLEDGES THAT IT AND THE OTHER PARTY HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 5.4.
- 5.5 Successors and Assigns. The rights of a Holder may only be assigned in accordance with Section 2.9 to a Permitted Transferee. A Permitted Transferee to whom rights are transferred pursuant to Section 2.9 may not again transfer those rights to any other Permitted Transferee, other than as provided in Section 2.9. Except as otherwise expressly provided herein, this Agreement shall be binding upon and inure to the benefit of the parties hereto and any and all successors to the Company and each Holder and their respective assigns.
- 5.6 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the parties hereto and their respective successors and permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever, under or by reason of this Agreement; provided, however, the parties hereto hereby acknowledge that the Persons set forth in Section 2.4(b) are express third-party beneficiaries in accordance with Section 2.4(b).
 - 5.7 Headings. The headings in this Agreement are for reference only and shall not affect the interpretation of this Agreement.
- 5.8 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced under any law or as a matter of public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties to this Agreement shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated by this Agreement be consummated as originally contemplated to the greatest extent possible.
- 5.9 Waivers. The observance of any term of this Agreement may be waived (either generally or in a particular instance and either retroactively or prospectively) by the party entitled to enforce such term, but such waiver shall be effective only if it is in a writing signed by the party against whom the existence of such waiver is asserted. Unless otherwise expressly provided in this Agreement, no delay or omission on the part of any party in exercising any right or privilege under this Agreement shall operate as a waiver thereof, nor shall any waiver on the part of any

party of any right or privilege under this Agreement operate as a waiver of any other right or privilege under this Agreement nor shall any single or partial exercise of any right or privilege preclude any other or further exercise thereof or the exercise of any other right or privilege under this Agreement. No failure by either party to take any action or assert any right or privilege hereunder shall be deemed to be a waiver of such right or privilege in the event of the continuation or repetition of the circumstances giving rise to such right unless expressly waived in writing by the party against whom the existence of such waiver is asserted.

- **5.10 Entire Agreement.** This Agreement, together with the Warrants and any related exhibits and schedules thereto, constitutes the sole and entire agreement of the parties to this Agreement with respect to the subject matter contained herein, and supersedes all prior and contemporaneous understandings and agreements, both written and oral, with respect to such subject matter. Notwithstanding anything contrary contained herein, in the event of any conflict or inconsistency between this agreement and Section 16(d) of the common stock purchase warrant in favor of the Initial Holder dated May 29, 2020, the terms of this agreement shall govern and control.
- **5.11** <u>Amendment</u>. This Agreement may not be amended or modified in any respect except by a written agreement signed by the Company and the Holders of a majority of the then-outstanding Registrable Securities.
- **5.12** Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when each party hereto shall have received counterparts hereof signed by each of the other parties hereto. If any signature is delivered by facsimile transmission or by PDF, such signature shall create a valid and binding obligation of the party executing (or on whose behalf the signature is executed) with the same force and effect as if such facsimile or PDF signature were an original thereof.
- **5.13** Further Assurances. Each of the parties to this Agreement shall, and shall cause their Affiliates to, execute and deliver such additional documents, instruments, conveyances and assurances and take such further actions as may be reasonably required to carry out the provisions hereof and to give effect to the transactions contemplated hereby.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

BAUDAX BIO, INC.

By: /s/ Gerri Henwood

Name: Gerri Henwood Title: Chief Executive Officer

MAM EAGLE LENDER, LLC

By: /s/ Lou Hanover

Name: Lou Hanover Title: Authorized Signatory

[Signature Page to Registration Rights Agreement]

Exhibit A

JOINDER AGREEMENT

Reference is made to the Registration Rights Agreement, dated as of March 29, 2023 (as amended from time to time, the <u>Registration Rights Agreement</u>"), by and among Baudax Bio, Inc., a company organized under the laws of the Commonwealth of Pennsylvania, and MAM Eagle Lender, LLC, and the other parties thereto, if any. The undersigned agrees, by execution hereof, to become a party to, and to be subject to the rights and obligations under the Registration Rights Agreement.

[NAME]			
By:			
Name: Title:			
Date:			
Address:			